

National Défense Defence nationale

DEFENCE RESEARCH AND DEVELOPMENT CANADA (DRDC) RECHERCHE ET DÉVELOPPEMENT POUR LA DÉFENSE CANADA (RDDC)



# A review of display and input technologies for the development of the Command Reconnaissance Area Coordination and Control Environmental Network (CRACCEN) decision aid system

Chrissy M. Chubala DRDC – Atlantic Research Centre

Terms of Release: This document is approved for public release.

**Defence Research and Development Canada** 

Scientific Report DRDC-RDDC-2022-R139 August 2022



**CAN UNCLASSIFIED** 

#### **IMPORTANT INFORMATIVE STATEMENTS**

This document was reviewed for Controlled Goods by Defence Research and Development Canada (DRDC) using the Schedule to the Defence Production Act.

Disclaimer: This publication was prepared by Defence Research and Development Canada an agency of the Department of National Defence. The information contained in this publication has been derived and determined through best practice and adherence to the highest standards of responsible conduct of scientific research. This information is intended for the use of the Department of National Defence, the Canadian Armed Forces ("Canada") and Public Safety partners and, as permitted, may be shared with academia, industry, Canada's allies, and the public ("Third Parties"). Any use by, or any reliance on or decisions made based on this publication by Third Parties, are done at their own risk and responsibility. Canada does not assume any liability for any damages or losses which may arise from any use of, or reliance on, the publication.

Endorsement statement: This publication has been peer-reviewed and published by the Editorial Office of Defence Research and Development Canada, an agency of the Department of National Defence of Canada. Inquiries can be sent to: Publications.DRDC-RDDC@drdc-rddc.gc.ca.

© Her Majesty the Queen in Right of Canada (Department of National Defence), 2022

© Sa Majesté la Reine en droit du Canada (ministère de la Défense nationale), 2022

CAN UNCLASSIFIED

# Abstract

The Command Reconnaissance Area Coordination and Control Environmental Network (CRACCEN) is a decision aid system being developed by Defence Research and Development Canada (DRDC) – Atlantic Research Centre to revolutionize the way the Royal Canadian Navy (RCN) conducts Underwater Warfare (UWW). This Scientific Report outlines the high-level goals and requirements of the CRACCEN system and reviews several fields of research in order to determine the appropriate candidate technologies for which to develop the system. A variety of potential display and input hardware technologies are evaluated with regard to both task requirements and characteristics of the UWW task environment. The results of this review suggest that a tabletop display capable of accommodating a group of simultaneous users be adopted as the central display for the CRACCEN system, with the possibility of using additional displays and display modes to augment the capabilities of the UWW team. Further research with Subject Matter Experts (SMEs) is recommended to validate the viability, usability, and added value of these displays, and to iteratively design interfaces that are well-suited to both the display type and the needs of the CRACCEN users.

# Significance to defence and security

The RCN has a long-term objective to revolutionize and modernize the conduct of UWW and the Command Reconnaissance Area Coordination and Control Environmental Network CRACCEN system will play a central role in this objective. In order for development and prototyping of the CRACCEN system to proceed, appropriate technologies need to be fitted to the demands of the tasks and the conditions of the task environment. This report reviews and analyzes various potential display and input technologies with regard to these requirements and provides concrete recommendations for prototype development and future research for eventual integration into CRACCEN.

# Résumé

Le Centre de recherche de l'Atlantique de Recherche et développement pour la défense Canada (RDDC) élabore actuellement un réseau environnemental de coordination et de contrôle de la zone de reconnaissance du commandement (CRACCEN). Il s'agit d'un système d'aide à la décision visant à révolutionner la façon dont la Marine royale canadienne (MRC) mène la guerre sous-marine (GSM). Ce rapport scientifique énonce les objectifs et les exigences de haut niveau du système CRACCEN et examine plusieurs domaines de recherche afin de proposer des technologies envisageables qui conviennent à son élaboration. Le rapport évalue une variété d'écrans d'affichage et de technologies matérielles d'acquisition de données en ce qui a trait aux exigences opérationnelles de même qu'à l'environnement opérationnel de la GSM. Les résultats de cet examen indiquent que l'affichage central choisi du système CRACCEN pourrait être sous forme de surface de table. Cet affichage permet d'accueillir plusieurs utilisateurs simultanément et offre a possibilité d'utiliser des écrans et des modes d'affichages supplémentaires afin d'accroître les capacités de l'équipe de la GSM. Il est recommandé de poursuivre les recherches avec des experts pour vérifier la viabilité, la convivialité et la valeur ajoutée de ces écrans, ainsi que pour concevoir des interfaces de manière itérative qui conviennent au type d'écran d'affichage et aux besoins des utilisateurs du système CRACCEN.

# Importance pour la défense et la sécurité

La MRC souhaite révolutionner et moderniser la conduite de la GSM, et le CRACCEN occupera une place centrale dans le cadre de cet objectif à long terme. Pour que le développement et le prototypage du système CRACCEN puissent aller de l'avant, les technologies appropriées doivent être adaptées aux exigences opérationnelles et aux conditions de l'environnement opérationnel. Ce rapport examine et analyse diverses technologies envisageables en matière d'affichage et d'acquisition de données en fonction de ces exigences et présente des recommandations concrètes concernant l'élaboration de prototypes et la recherche future aux fins d'une intégration ultérieure au système CRACCEN.

# Table of contents

Abstract		i		
Significance to defence and security				
Résumé				
Importance pour la défense et la sécurité				
Table of contents	i	iii		
List of tables.		v		
Acknowledgements		vi		
1 Introduction $\ldots$		1		
1.1 Report outline		1		
2 CRACCEN		2		
2.1 The CRACCEN project		2		
2.2 The CRACCEN system		2		
2.3 System requirements for CRACCEN		2		
2.3.1 Deriving requirements.		2		
2.3.2 Using requirements to constrain display and input technology selection .		3		
2.4 System users		4		
3 Method		5		
3.1 Databases		5		
3.2 Search parameters		5		
3.3 Review topics		6		
3.3.1 Review of display and input technologies		6		
3.3.2 Suitability of display and input technologies for teamwork		6		
3.3.3 Suitability of display and input technologies for the UWW task environment		7		
4 Review of display and input technologies		8		
4.1 2D displays		8		
4.2 3D displays		8		
4.3 Extended reality displays		9		
<ul><li>4.3.1 Virtual reality HMDs</li></ul>		9 9		
4.3.2 Augmented rearry minds		9		
4.3.4 Spatial AR		0		
5 Review of technology suitability		1		
5.1 Suitability of display and input technologies for teamwork		1		
5.1.1 2D central display		1		
5.1.1.1 Input methods for 2D displays		2		
5.1.1.2 Text input for 2D displays		3		
5.1.2 Incorporating 3D information		4		
5.1.2.1 Stereoscopic and auto-stereoscopic 3D displays		4		
5.1.2.2 Extended reality displays		5		
5.1.2.3 Input methods for 3D displays and extended reality	1	6		

		5.1.3	Combining multiple displays	17
	5.2	Suitab	ility of display and input technologies for the UWW task environment	18
		5.2.1	Eye strain from display technologies	18
		5.2.2	Cybersickness from extended reality HMDs	19
		5.2.3	Impacts of ship motion on input technologies	20
		5.2.4	Impacts of ship vibration on display and input technologies	20
		5.2.5	Ergonomics of display and input technologies	21
6	Reco	mmend	lations	23
	6.1	Large	central tabletop 2D display	23
	6.2		le display angle	23
	6.3		al AR HMDs and devices for individual work	24
	6.4	-	eresearch	24
		6.4.1	Ideation studies for flexible displays	25
		6.4.2	Testing prototypes for the central display.	25
		6.4.3	Testing AR HMD and individual work prototypes	26
7	Conc	lusions		27
Re	ferenc	es		28
An	nex A	Integ	rative review summary	53
		-	abbreviations/acronyms/initialisms	56

# List of tables

Table 1:	A list of search terms used to review the suitability of display and input technologies for teamwork.	6
Table 2:	A list of supplemental search terms used to review the suitability of display and input technologies for the UWW task environment.	7
Table A.1:	A summary of the findings from the integrative review, assessing the extent to which a given display or input technology (rows) satisfies the criteria of interest for CRACCEN (columns).	53

# Acknowledgements

The author would like to thank Tania Randall for her assistance with proofreading, editing, and proving general guidance over the development of this Scientific Report. Additional thanks is owed to an anonymous reviewer, whose feedback greatly shaped and improved the final form of this report.

# 1 Introduction

The Royal Canadian Navy (RCN) is invested in revolutionizing and modernizing the conduct of Underwater Warfare (UWW). The Command Reconnaissance Area Coordination and Control Environmental Network (CRACCEN) project is a first step toward reaching this objective. Rather than providing incremental improvements to individual elements of the UWW system, the project aims to rethink existing Command and Control (C2) structures. Whereas the potential to supplant existing systems and structures with revolutionary new ones provides invaluable opportunities to streamline and improve UWW conduct, without being constrained by current systems, the open-ended nature of the project introduces new complexities.

Namely, the question of what such a revolutionary system should look like is left entirely open. Any number of existing, emerging, or future technologies for display and interaction might be applied to the problem to varying degrees of effectiveness. For development and prototyping of the CRACCEN system to proceed, the design space must be narrowed, and appropriate technologies must be selected. This Scientific Report reviews available display and input technologies with regard to the requirements of the tasks to be completed with the CRACCEN system and offers concrete recommendations for technology selection and future research.

# 1.1 Report outline

This report is organized as follows. Section 2 describes the CRACCEN project and system, and leverages previous work to define the system requirements that set the stage for the subsequent review and analysis. Section 3 discusses the methodological approach taken for this integrative review. Section 4 reviews the technologies that might be applied to the CRACCEN system. Section 5 evaluates the suitability of the technologies with regard to two types of constraints: the characteristics of the tasks to be accomplished by the UWW team using the CRACCEN system, and the conditions of the shipboard UWW task environment. Section 6 draws recommendations for display and input technology selection from the results of the integrative review, and outlines the future research required to validate and build upon the recommendations. Section 7 concludes the discussion of display and input technology selection for the CRACCEN system.

# 2 CRACCEN

Under an extended peacetime, the conduct of UWW within the RCN has required minimal modernization. By contrast, the same period has seen dramatic shifts in the technological landscape, and thus in the environment within which UWW is conducted. In 2017, Defence Research and Development Canada (DRDC) – Atlantic Research Centre responded to a Direct Client Support task issued by the Director Science & Technology – Navy for a system that will modernize and revolutionize the way the RCN plans and conducts UWW [1]. DRDC – Atlantic Research Centre's proposal encompasses a five-year project of research and development for a system to overhaul existing methods and procedures for predictive situational awareness, battlespace management, operational planning, and mission execution [2].

# 2.1 The CRACCEN project

An earlier internal publication breaks the CRACCEN project down into actionable Work Breakdown Elements (WBEs), covering the wide range of efforts required to revolutionize UWW conduct [2]. This report situates itself within WBE 1: Design Concepts for Revolutionary C2 Systems, the objectives of which are to develop design concepts and workflows for CRACCEN and to determine the best methods for visualization. Development of the CRACCEN system is expected to extend well beyond the five-year timeframe of this project, which aims only to lay the foundations for a revolutionized UWW system. For the purposes of this report, references to CRACCEN hereafter refer to the intended future system rather than the five-year project.

# 2.2 The CRACCEN system

The CRACCEN system is intended to support UWW command teams with automated information collation, environmental prediction, real-time modelling and simulation tools, and novel decision support tools [2]. Current planning and decision-making structures available to UWW command teams see team members operating stove-piped and individual-focused desktop software, collating data manually in order to develop situation awareness (SA), and compiling their own visual aids for course of action (COA) development and mission planning [3]. By contrast, the CRACCEN system is conceptualized to be a central collaborative environment that provides command teams with all the information relevant to their mission planning and decision-making, and that supports team discussions, collaborative COA development, and the establishment and maintenance of shared SA.

The primary focus of this report is CRACCEN's collaborative interface(s) to be used by command teams. As described in the Naval Order for Directed Client Support, CRACCEN should involve a collaborative interface capable of displaying dynamic (e.g., predictions of oceanographic conditions, environmental risk assessments, underwater threat tracks) and static (e.g., charted wrecks, marine mammal sightings, historical bathymetry) information layers to enable a central command and control environment for UWW mission planning and execution [4]. Beyond the scope of this report, the system will more broadly involve an extensive information backbone and a suite of algorithms to support the automated collection, collation, processing, analysis, sharing, and dynamic visualization of these data [2].

# 2.3 System requirements for CRACCEN

### 2.3.1 Deriving requirements

A number of semi-formal and descriptive analyses can be employed to characterize the nature of the C2 environment, with the goal of improving a C2 work system. Hierarchical Goal Analysis (HGA) identifies the

goals that a work system needs to achieve in order to support its operators [5]. Because HGA focuses on goals, which may be nonlinear or parallel, rather than on the linear stream of tasks by which the goals are accomplished using current systems, it is an ideal analytical approach to deriving requirements for the design of new or revolutionary systems such as CRACCEN [6].

Past work has explored the nature of the UWW C2 environment in the RCN using HGA [3][7][8][9]. A recent report specifically identified four high-level goals for consideration in the design of CRACCEN [10]. Requirements for CRACCEN to achieve these goals are described below:

- 1. Assessing Readiness: CRACCEN should automatically collect and collate information about the status (e.g., location, bearing, weapons status, sensor status, and positioning) of own ship and task group (TG) assets, and provide that information to the UWW team, upon command, in a spatial visualization medium that is easy to parse and intuitive to interact with.
- 2. *Planning and Managing*: CRACCEN should facilitate both short- and long-term planning exercises by a) automatically collecting and collating data from multiple data streams to assist in building the recognized maritime picture; b) using artificial intelligence (AI) to predict future states of data streams and to produce planning recommendations; and c) providing tools to help the UWW team collaboratively build, visualize, and promulgate COAs and plans.
- 3. *Maintaining SA*: CRACCEN should streamline and automate the development of SA, such that UWW teams are provided a real-time overview of the area of operations (AOO) on a central display. This will decrease the likelihood that pertinent information is overlooked by minimizing the cognitive burden, and will allow teams to focus on the mission planning and execution tasks. A central CRACCEN display showing information pertinent to SA will also allow SA to be shared between members of the UWW team, thereby facilitating the collaborative development of COAs.
- 4. *Operational Tasks*: CRACCEN should facilitate mission execution by a) incorporating multiple sources of information (e.g., bathymetry, tactics, historical data on the behaviour of sighted enemies, intelligence) into algorithms that predict areas of high risk for enemy submarine activity; b) using AI to predict future locations of previously sighted submarines; and c) producing recommendations for sensor and asset placement in order to best protect the TG from threats.

Overlaps and interdependencies exist between these high-level goals, and together they elucidate the intended and expected use cases for CRACCEN. The focus of this report is on the display and input technologies that can support these requirements.

### 2.3.2 Using requirements to constrain display and input technology selection

Preliminary constraints for technology selection can be established on the basis of the requirements defined above. In order to meet the requirements of CRACCEN, at least one central display will be required; the exact nature of and methods of interacting with this central display are open to exploration. Because UWW necessarily takes place in a three-dimensional (3D) space, from the surface of the water down to the sea floor, considerations for 3D display capabilities must also be made in order to best support SA. Several methods for 3D display will be explored in this report to determine their feasibility for integration with CRACCEN. Finally, individual UWW team members are experts in their own domains and roles, and will need to complete ongoing individual work alongside the collaborative work facilitated by CRACCEN. Indeed, the collaborative work of mission planning and execution depends upon the individual expertise brought to the problem. As such, combinations of multiple displays, for a flexible blend of individual and group work, should also be considered.

Although the ability to collaborate remotely with higher command or other TG members may be a desired capability in the future, the present work focuses solely on the co-located collaborative work that will serve as CRACCEN's central purpose.

# 2.4 System users

Mission analyses, goal decomposition analyses, and task analyses have been conducted to identify the personnel expected to use CRACCEN, corroborated by interviews with subject matter experts (SMEs) [10][11]. Personnel expected to use the system include the Anti-Submarine Plot Officer (ASPO), Anti-Submarine Warfare Commander (ASWC), Commanding Officer (CO), Current Operations Officer (COpsO), Future Operations Officer (FOpsO), Force Track Coordinator Subsurface (FTC-SS), Operations Room Officer (ORO), Sonar Control Supervisor (SCS), and Underwater Warfare Director (UWWD). Together, these users comprise the UWW command team, though it is unlikely that all of these roles will be involved in planning and decision-making at once. CRACCEN should be able to accommodate three to five simultaneous co-located users, with the flexibility to accommodate more as needed.

# 3 Method

The central research question of this report can be summarized as "which display and input technologies are best suited to the anticipated use cases of a revolutionary decision aid for UWW command teams?" Even in this most simplified form the question cuts across several disciplines, as it must consider display and input technologies, characteristics of the tasks to be completed, and environmental and other pragmatic constraints. Some of these disciplines include C2, engineering, ergonomics, human-computer interaction, human factors, team cognition, and psychology. As such, the methodological approach to this report follows that of an integrative review, which aims to generate new frameworks and perspectives on a topic by reviewing and synthesizing existing literature [12]. This approach to literature review is particularly useful when a research question is too broad or interdisciplinary for a fully systematic review.

The research question (i.e., which display and input technologies are best suited to the anticipated use cases of a revolutionary decision aid for UWW command teams?) can be broken down into three distinct parts necessitating different approaches to literature search and review: (1) a description of potential technologies; (2) the suitability of the technologies for teamwork and collaboration; and (3) the suitability of the technologies for the shipboard environmental conditions under which UWW teams will be completing their tasks. Section 5 synthesizes results from all three aspects of the review and may be read alongside the summary table of the results provided in Annex A.

# 3.1 Databases

Databases consulted for the integrative review were: Institute of Electrical and Electronics Engineers (IEEE) Xplore, Inspec, ScienceDirect, Canadian Defence Information Database (CANDID), Defense Technical Information Center (DTIC), and Google Scholar.

# 3.2 Search parameters

For all literature searches described below, results were limited to those written in English and those published between 2010 and 2022, in order to ensure relevance of the findings to current and future requirements. Results which were not from peer-reviewed journal articles, book chapters, conference proceedings, or military reports were discarded from consideration. In line with published literature on best practices, searches of the Google Scholar database were limited to only the first 100 results, sorted by relevance [13]. Searches of the DTIC technical report database were likewise limited because this database is powered by Google and does not show more than the first 100 results, sorted by relevance, even when the search returns more than 100 results. For all other databases, no such limitations were applied.

For all searches, keywords of interest had to be included in the Abstract or Document Title<sup>1</sup>. Titles and abstracts of all results were read, assessed for relevance, and discarded if deemed irrelevant; for example, results yielded in the searches regarding the suitability of a technology for teamwork were discarded if they did not, in fact, pertain to team activities. Retained items were read fully and further screened for relevance. In the course of reading, reference lists were scanned for additional relevant references that had not been identified by the database search. These secondary sources were flagged and assessed in a similar manner; year of publication was not considered for these secondary sources, as they often provided background information or more foundational evidence.

<sup>&</sup>lt;sup>1</sup> With the exception of DTIC and Google Scholar, for which Abstracts could not be individually specified as search parameters; instead, the full text was searched for these databases.

In some instances, an article uncovered in a database search revealed a conference proceeding for a specialized topic of direct relevance to the research question (e.g., IEEE International Workshop on Horizontal Interactive Human-Computer Systems). In these instances, the conference proceedings were hand-searched for additional relevant results.

# 3.3 Review topics

### 3.3.1 Review of display and input technologies

For this aspect of the review, the field of potential technologies was derived from the system requirements discussed in Section 2. Namely, CRACCEN will require at least one central display and the possibility of augmenting the display(s) into three dimensions. The results of the review of technologies can be found in Section 4.

Descriptions of display technologies were developed from existing review papers, which served to define the space of possible technologies. No review paper was consulted to describe two-dimensional (2D) displays, as the technology is well-established and most meaningful differences emerge in later considerations of input methods and task demands. Two review papers provided the foundation a review of 3D display technologies [14][15]. Additional review papers were consulted for the discussion of extended reality, which includes augmented reality (AR) and virtual reality (VR) head-mounted displays (HMDs), as well as AR with handheld devices [16][17].

### 3.3.2 Suitability of display and input technologies for teamwork

For this aspect of the review, searches were made for individual technologies or categories of technologies with regard to teamwork specifically, using the keywords "teamwork," "team," "collaboration," and "group work." Additional keywords selected for each technology are listed in Table 1. The results of this portion of the review are collected for discussion in Section 5.1.

Торіс	Keywords
Display size	display size, screen size, large display
Display orientation	tabletop, horizontal display, horizontal screen, vertical display, vertical screen, display orientation, screen orientation
Mouse input	mouse input, computer mouse, multiple mouse
Pen input	stylus, styli, pen input, multiple pen, pen touch
Touch input	touchscreen, touch screen, touch input, multitouch
Trackball input	trackball <sup>2</sup>
Text input	text input, keyboard, speech recognition, handwriting <sup>3</sup>
3D displays	3d glasses, eyeglasses, 3d display, parallax display
Extended reality	virtual reality, augmented reality, extended reality, co-located, co-location

Table 1: A list of search terms used	to review the suitability of display an	d input technologies for teamwork.

The scope of this review focused on co-located collaboration of real humans, and results pertaining to remote collaboration and human-robot collaboration were thus discarded. A large proportion of the search results for

<sup>&</sup>lt;sup>2</sup> No relevant results were returned for the team-related search with this keyword.

<sup>&</sup>lt;sup>3</sup> No relevant results were returned for the team-related search with this keyword.

extended reality technologies pertained to remote and/or networked collaboration setups. The additional keywords "co-located OR co-location" were thus added to this particular search in order to exclude these findings and better focus on results of interest.

### 3.3.3 Suitability of display and input technologies for the UWW task environment

For this aspect of the review, searches were made for ergonomic properties of individual technologies or categories of technologies, using the keyword "ergonomics." Additional keywords employed for each technology or category of technologies is the same as those given in Table 1. Note that for these searches, the inclusion range for publication year was broadened to 2000–2022 in order to capture more results.

These searches were supplemented with targeted searches for the topics of eye strain and cybersickness, which are known issues particularly for VR and AR HMDs. Note that for these targeted searches, the author focused on experiments comparing technologies and review studies, rather than the wide range of experimental studies testing the possible etiologies or exploring real-time detection and prediction of the phenomena. Finally, targeted searches were made for the impacts on display and input technologies of known environmental conditions of the shipboard UWW task environment: ship motion and vibration.

The keywords for these supplemental searches are given in Table 2. The results of this portion of the review are collected for discussion in Section 5.2.

Торіс	Keywords
Eye strain	eye strain, visual fatigue, virtual reality, augmented reality, hmd, head-mounted display
Cybersickness	cybersickness, simulator sickness, virtual reality, augmented reality, hmd, head-mounted display
Ship motion and vibration	ship motion, physical motion, engine vibration, motor skills, manual control, input device, text input, virtual reality, augmented reality, human factors

Table 2: A list of supplemental search terms used to review the suitability of
display and input technologies for the UWW task environment.

# 4 Review of display and input technologies

This review outlines display and input technologies that may be considered for CRACCEN. This section is not intended as an in-depth review of all possible display and input technologies, but rather provides a brief overview of the many technologies that at least approximately meet the system requirements previously identified, so that their suitability for CRACCEN can be more carefully explored in Section 5.

# 4.1 2D displays

2D displays are familiar and ubiquitous technologies, available in a wide range of sizes. For CRACCEN, the target size of a central 2D display is one large enough to accommodate multiple users, but not so large as to be impractical for installation aboard a naval vessel. The orientation of 2D displays can likewise vary, from a vertical orientation to a horizontal orientation, or tilted at an angle between vertical and horizontal.

The size and orientation of a large display have bearing on a number of design constraints, which will be considered in Section 5, including how many users can concurrently view and interact with the display, the viability of various input methods, and the display ergonomics. 2D displays in a horizontal orientation are typically referred to as tabletop displays when they are targeted toward multiple users, as they allow users to stand or sit around a table-sized horizontal display and look down at the information from above. Tabletop displays are often used to digitally recreate the experience of collaborative work around a physical tabletop. The term "tabletop display" will be used throughout this report to refer to such horizontally-oriented 2D displays.

# 4.2 3D displays

The UWW arena, stretching from the water's surface to the sea floor, is three-dimensional. Therefore, there are inherent limitations on the ability of 2D displays to accurately represent the AOO. UWW teams currently conduct their operations with a variety of 2D graphical systems, but the development of CRACCEN represents an opportunity to revolutionize and simplify the workflow. Although a 3D space can be inferred from the combination and/or manipulation of 2D views, providing an intuitive 3D view may lighten the cognitive load on UWW teams and free additional mental resources for the complex mission planning and execution tasks required of them [18]. A brief overview of 3D display technologies is provided below, as collated from two recent review papers [14][15]. Whereas many 3D displays have been developed for display purposes only (i.e., no consideration for user interaction) the focus below is only on those 3D displays for which interactive mechanisms have been developed, in order to meet CRACCEN's requirements.

Perhaps the most familiar type of 3D displays are wavelength selective stereoscopic systems, which present two slightly offset and colour-shifted images to the user. The user wears special stereoscopic glasses that filter information such that only certain wavelengths (i.e., colours) reach each eye, and the effect when the two images are combined by the brain is a 3D view [19]. Similar systems use polarizing lenses to filter out a different image to each eye, with each image presented either simultaneously [20] or in very rapid succession [21].

Auto-stereoscopic (AS) systems allow for the stereoscopic viewing of 3D scenes without any special glasses by presenting slightly different information to each eye in order to invoke the illusion of depth. These systems can be time-sequential, presenting light from two different directions (i.e., each toward one eye) in rapid alternating sequence [22]; or time-parallel, with a layer between the viewer and the display that simultaneously renders a slightly different picture to each eye. Although AS eliminates the need for specialized wearable devices, it often necessitates other highly specialized hardware (e.g., [23][24]). Other AS 3D display systems produce multiple different, slightly offset images, each of which is only visible within a limited viewing zone and thus only visible

to one eye at a time [25][26][27][28]. When the user moves her head, she moves through viewing zones and thus changes the view of the 3D object. Finally, volumetric 3D displays present imagery in true 3D space by illuminating volumetric pixels suspended in midair, usually within an enclosed surface [29]. Volumetric displays require specialized hardware (i.e., the display volume), and incur a high computational cost.

# 4.3 Extended reality displays

Extended reality is an umbrella term that encompasses the combination of virtual and real environments, with a strong emphasis on immersion and interaction: extended reality technologies allow a user to not only view virtual 3D information, but to interact with real or virtual objects while moving through real or virtual environments<sup>4</sup>.

Per the taxonomy of [15], virtual reality (VR) and augmented reality (AR) head-mounted 3D displays (HMDs) are two-view stereoscopic displays employing advanced stereoscopic glasses. However, they are somewhat special among such technologies in that they afford a full parallax experience, wherein the 3D view changes with both horizontal and vertical head movements. As a result, they have garnered much more attention and research effort in recent years than have non-immersive 3D displays. Each is considered individually below, alongside handheld AR and spatial AR.

### 4.3.1 Virtual reality HMDs

VR HMDs project a virtually rendered environment onto small optical displays mounted in front of the user's eyes for a fully immersive experience [14][16]. From within the HMD, users cannot see the external world, including their own bodies and hands, unless objects in the external world are tracked and reproduced inside the virtual world. Users are therefore often given camera-tracked handheld remotes in order to interact with the virtual world. VR HMDs typically present auditory information to users through speakers positioned directly on the HMD.

Some commonly used commercial VR HMD systems today are the HTC Vive (HTC) and the Oculus Rift (Meta Technologies, LLC). Both systems currently require the HMD, and by extension the user's head, to be physically tethered to a powerful computer and it is recommended that only one headset be tethered per computer in order to ensure low latency of the displayed virtual environment. Both systems also employ handheld controllers to allow users to navigate the virtual environment and interact with virtual objects. Some VR systems employ headset-mounted cameras whereas others require a set of external cameras in order to track headset and controller motion. External cameras must be carefully calibrated every time they are moved, so it is generally recommended that they remain in one place after calibration. Other consumer-level VR HMD systems have been developed to function without the use of peripheral cameras or tethering (e.g., Oculus Go, Oculus Quest, Lenovo Mirage).

### 4.3.2 Augmented reality HMDs

AR HMDs are devices that overlay virtually rendered elements onto the user's view of the real world [14][16]. Forward-facing cameras track the 3D structure of the real-world environment and user-facing sensors track head and/or eye movements, allowing users to view 3D objects from all angles by moving around in physical space. Forward-facing cameras can also be used to track the user's hand movements, thereby allowing them to manipulate virtual objects. Like VR HMDs, AR HMDs typically present auditory information directly to users through speakers positioned near the ears on the HMD.

<sup>&</sup>lt;sup>4</sup> Although they are considered a form of extended reality, Cave Automatic Virtual Environments (CAVEs) immerse the user by projecting an interactive environment onto three to six walls of a room. These solutions are not viable within a shipboard setting, and are thus not considered in this report.

Like VR HMDs, AR HMDs can involve a full video screen in front of the user's eyes, whereupon a video feed of the real world is presented and overlaid with the virtual objects. More commonly, however, AR HMD screens are transparent, allowing a direct view of the real world that is then overlaid with virtual objects within a certain visual range. The Microsoft Hololens2 is presently the most commonly used example of a see-through AR HMD.

### 4.3.3 AR with handheld devices

Instead of wearing a HMD, users can experience video-feed AR by holding up handheld devices (e.g., smartphones, tablets) as windows onto the virtually augmented environment [16][17]. Forward-facing cameras in the devices sense the real 3D environment and present a live video feed onto the handheld display alongside virtually rendered objects. Accelerometer and gyrometer data from the device help the system track movement and accurately render the 3D objects as the user (or device) moves around in physical space, and the device's touchscreen allows users to interact with virtual objects displayed upon it.

### 4.3.4 Spatial AR

Spatial AR, also known as tangible AR, uses a system of cameras and projectors to track physical objects in the environment and project virtual information onto them. This allows users to view and interact tangibly with physical objects that are digitally augmented with additional visual information, without the need for glasses, HMDs, or specialized displays [17][30]. The technique can be used to display complementary text annotations on a physical object, to demonstrate texture on an un-textured object, to provide users with a virtual view into the 3D interior of an object, or to transform a non-digital object (e.g., a piece of paper) into a digital display. Common spatial AR systems employ a tabletop sandbox, within which users can shift sand to deform the physical landscape and onto which environmental details can be optically projected<sup>5</sup>.

Spatial AR is particularly useful for architectural and design fields (e.g., being able to physically interact with a quick digital mock-up of new products), as well as for some training purposes (e.g., being able to visualize 3D organs projected onto a physical dummy for surgical training). However, its usefulness to the UWW battlespace is debatable, because the desired target for interaction is the entire surface and subsurface of the AOO. Although projecting the UWW battlespace onto a large physical 3D box could aid with general SA, user interactions would be limited by the external dimensions of the physical box. The degrees of freedom for manipulation would be insufficient to support exploration of and interaction with that 3D space for mission planning and execution. Spatial AR is thus not considered any further in this report.

<sup>&</sup>lt;sup>5</sup> Physical sand tables are increasingly being superseded by virtual sandboxes facilitated by AR or VR, especially in the military domain [31].

# 5 Review of technology suitability

As described in Section 2, the requirements for CRACCEN recommend at least one central 2D team display for planning and decision-making purposes, a method for incorporating interactive 3D information, and the possibility of incorporating multiple displays. The primary consideration in this report is the suitability of display and input technologies for teamwork and collaboration. A secondary consideration is the suitability of each technology to the special constraints of the shipboard UWW task environment. The suitability of technologies for individual work is considered as a baseline throughout, particularly as it pertains to the selection of input methodologies and ergonomics, which have not often been studied in collaborative settings. A table summarizing the findings is presented in Annex A, which facilitates direct comparisons between technologies.

# 5.1 Suitability of display and input technologies for teamwork

### 5.1.1 2D central display

The number of simultaneous CRACCEN users may provide an important constraint on the appropriate 2D central display technologies. Results from earlier task, mission, and work analyses suggest that CRACCEN should be able to accommodate a group of three to five co-located SMEs, with room for further users as needed [10][11]. Because multiple users are anticipated to interact with CRACCEN simultaneously, larger displays (i.e., greater than 32") should be considered as the central display technology for CRACCEN. A recent review of large displays found distinct advantages over more traditional individual displays (e.g., desktops, laptops) in facilitating communication, coordination, and workspace awareness [32]. Research comparing "small" (17") and "large" (33") displays directly has shown that larger displays more readily support an equitable distribution of collaborative task activities than do smaller displays [33] (c.f. [34], which found increases in the quantity of communication with smaller central displays, likely as a compensation for the smaller and less effective information display). Handheld 2D display devices on their own are not ideal for collaborative work, as they discourage communication, decrease workload equity, and lower the quality of solutions, relative to larger shared workspaces [35]. Space constraints aboard a naval vessel generally provide a practical upper limit on the size of a central 2D display, though previous work has shown full-wall (e.g., 120") displays can improve shared SA in military command teams at land-based command centres [36][37].

Screen display orientations include vertical, horizontal (e.g., tabletop), or angled displays that fall between vertical and tabletop orientations. Tasks that will require users to spend a large portion of their time interacting with the system may benefit more from vertical displays, whereas tasks requiring a lot of interpersonal interaction, including group situational understanding, discussion, and non-parallel work, may benefit more from tabletop displays [38]. Indeed, tabletop displays are typically reported to naturally and comfortably support efficient collaboration and face-to-face communication, by leveraging the familiar schemas of collaborating around a physical table and incorporating naturalistic communication modes to facilitate knowledge transfer [33][39][40][41][42][43][44]. The ability to work around any edge of the screen also allows for partitioning of individual work spaces that can later be combined for group-level activities [45][46]. Feedback notifications about other users' actions may reduce work redundancies and increase SA during such parallel tasks [47]. By contrast, communication around a vertical screen is less natural and more asymmetrical (e.g., one person talking, others listening), and is typically associated with decreased role switching, idea exploration, and other-awareness [39] (c.f. [48]).

Whereas vertical displays are better suited to shorter and more focused tasks with fewer users, because the vertical orientation limits users' ability to interact with the screen without bumping into one another, tabletop displays are better suited to tasks of longer duration that require more discussion [33]. Some research suggests that a full-wall display shared between few collaborators does not carry the same limitations on collaborative

work as smaller vertical displays [49], likely due to the increased surface area and space for communication. Less research has examined tilted screens that fall between vertical and tabletop orientations. Although initial evidence suggests a tilted screen may be preferred over vertical and tabletop screens for pairs of collaborating users [50], it is not clear whether the preference predicts better performance, nor whether the preference would hold for collaborations between more than two users.

Despite the apparent benefits of tabletop screens for collaborative work, it must be noted that distributing users around all edges of a tabletop screen introduces complications with the orientation of displayed elements [51][52]. Considerations for orientation would need to be made during user interface design for a tabletop display in order to ensure that multiple users can effectively comprehend, communicate about, and coordinate on the information presented [53]. The orientation of text information presents a particular challenge for tabletop displays, because the orientation of text relative to a reader has a significant impact on reading speed and accuracy. However, solutions exist for rotating and translating digital objects on horizontal touch displays, whether manually or automatically via sensors that detect users [54][55]. User orientation around a tabletop display is less of a concern when the primary information to be presented is spatial or graphical rather than textual. Notably, however, research suggests that groups tend to cluster around one edge of a tabletop display in order to share a common perspective of a 2D map [56], because users find spatial navigation disorienting when they view a map from different angles [57].

Flexibility may be built into the CRACCEN system by allowing the central 2D display to be convertible between vertical and tabletop orientations, to accommodate different task requirements [58].

#### 5.1.1.1 Input methods for 2D displays

Input methods refer to the technologies and mechanisms that allow users to interact with information presented on a display. Input methods can involve direct mapping between the user's actions and the display (e.g., direct touch screens, pen styli<sup>6</sup>) or indirect mappings, whereby the user's movements are captured by some device that translates to movement on the display (e.g., mouse, trackball).

Comparing direct and indirect input methods for individual users, direct touch input is faster, more accurate, and preferred relative to mouse input when the task is bimanual. By contrast, all metrics generally favour the mouse in most studies of single point interactions (e.g., [59][60]), except when the task involves drawing or games [61]. [62] found increased speed but decreased accuracy for single touch inputs relative to mouse and stylus inputs. For tasks requiring a high degree of precision, mouse input is generally preferred to touch input [59][62][63], except when manipulation of 3D information is required [64]. When high precision is not required for a task, mouse input is slower than direct touch input [65].

The familiarity of mouse input is likely a contributing factor in many of the comparisons above, as touch input accuracy appears to improve with experience [66]. Indeed, more recent studies indicate little difference between touch and mouse input for some tasks, perhaps reflecting the public's growing familiarity with touchscreen technologies since the commercial proliferation of handheld touch devices [67]. A variety of software-based adjustments to direct touch input show promise for overcoming its inaccuracies for small targets [68]. Pen styli can also offer improved precision because their point of contact is smaller and produces less occlusion and incidental touches than a finger [62][69][70], but the trade-off may be reduced speed compared to finger touch [71]. The incorporation of visual feedback when hovering a tracked pen stylus over a surface (e.g., comparable to a mouse cursor) can offer slight improvements to speed and dramatic improvements to accuracy of stylus inputs [72].

Trackball movement is both slower and less accurate than traditional mouse movements for most tasks, due to the difficulty of translating the rotary movement of a trackball to the linear space of a 2D display [73]. On larger

<sup>&</sup>lt;sup>6</sup> A pen stylus is a pen-shaped tool used for input on touch devices. It provides users with a fine tip for higher precision inputs than can generally be achieved with a finger.

displays, keeping track of a cursor for indirect input mappings can prove challenging, particularly when mouse or trackball acceleration is set to high [74]. When input device acceleration is set to low, it can become tedious to move a cursor across a large display. There is some evidence that trackballs have an advantage when moving across a large display at a high velocity, but users still suffer from decreases in speed and accuracy when slowing down to select a target [75].

Direct touch input carries its own drawbacks for larger displays. For example, users find dragging and object rotation tasks more difficult as the display size increases [76]. A user's ability to interact with information on the screen is also necessarily limited by the extent of her reach, which is in turn limited by the display orientation and whether she is seated or standing [77]. External tracking cameras can be employed with gesture-based inputs to extend reach on large displays [78], or to identify and differentiate individual user inputs [79]. Pen styli can also be used to differentiate simultaneous inputs from multiple users.

Finally, the relative nature of mouse and trackball inputs become less intuitive on tabletop displays, where users can be situated on any edge of the screen such that the relative motion of the cursor may not correspond to the direction the mouse or trackball is moved from the user's perspective. Critically for the purposes of CRACCEN's multi-user setting, it is difficult for users to locate one another's cursors and track movements when multiple indirect inputs are used [38][80][81][82].

Multiple mice may also decrease other-awareness and collaboration in favour of divide-and-conquer strategies, compared to single-mouse setups [83] (c.f. [84], which shows a potential role for multiple mice in mixed-focus collaborative tasks on wall-sized displays). By contrast, direct touch input improves other-awareness in collaborative tasks, because of the ease with which others' interactions with the display can be seen [82][85][86][87]. Likewise, both verbal and non-verbal communication improve when teams use individual tablet computers compared to individual small netbook laptops with keyboards [88], suggesting that the benefit of direct touch input for teamwork may hold true regardless of screen size. Indeed, direct touch and pen inputs provide better supports for interpersonal interaction around collaborative displays than do indirect (e.g., mouse) inputs, because they allow for gesturing, more effortless awareness of other users' actions, and more naturalistic inferences about intentions [80][81] (c.f. [89], which showed no difference in collaborative task performance or individual preferences between mouse and touch inputs to a tabletop display). Direct multi-touch around a tabletop is particularly conducive to communication and equitable collaboration, compared to single touch, single mouse, and multiple mouse inputs [90]. Currently available touchscreens support a high level of multi-touch (e.g., upwards of 30 simultaneous touch inputs), allowing multiple users to interact intuitively and simultaneously with a display, with or without the aid of pen styli.

### 5.1.1.2 Text input for 2D displays

Although the various task analyses that have been conducted on command team members suggest that CRACCEN usage will not involve a substantial amount of text input [10], it may nevertheless be necessary for users to enter data, annotate information, or type short notes to one another.

Generally speaking, physical keyboards provide the most familiar method of text entry, and thus produce the fastest and most accurate results without training. However, multi-user interactions with the display would require either multiple physical keyboards, which take up space and clutter the workstation, or a single keyboard that can be used in turn. For vertically oriented displays, the placement of a physical keyboard is self-evident, as it makes ergonomic sense to have keyboards placed in front of vertical displays. For tabletop displays, however, the appropriate placement of physical keyboards is not immediately apparent, since users can interact with a tabletop display around any edge.

An alternative to physical keyboards is digital touch or soft keyboards, which can be used with any touch display. Unlike physical keyboards, soft keyboards can be rotated, moved, and resized to accommodate user needs around a vertical or tabletop display, and multiple soft keyboards can be available for simultaneous text

#### DRDC-RDDC-2022-R139

entry tasks. Soft keyboards may thus be better suited than physical keyboards to collaborative work on large displays, although methods to improve speed and accuracy should be considered during the design phase. Usability studies, for example, have found that soft keyboards yield lower speed and accuracy than physical keyboards, likely due to the absence of haptic feedback from physical keys [91][92]. Dynamic resizing of keys based on letter-prediction algorithms has shown some promise in increasing speed and accuracy [93], as has adaptive re-spacing of keys in response to personal ergonomic habits [94][95]. Care should be taken to minimize input latency and, where possible, introduce other modalities of keystroke feedback (e.g., sound, vibration) in order to improve performance and usability [96][97].

Soft keyboards can also be activated by pen stylus touch, although this limits input speed greatly relative to the use of all ten fingers. A gesture-based solution allows for users to type without lifting a finger or stylus as they move through visually presented letters [98]. However, because these keyboards present characters in layouts designed to increase gesture fluidity (i.e., circular layouts), they are unfamiliar and therefore difficult to use [99].

Natural handwriting on touch surfaces with either pen styli or fingers allows for an intuitive text entry method and handwriting recognition algorithms are now sufficiently advanced to provide fast and accurate input processing. Notably, speed and accuracy of handwriting are much higher with pen styli with than fingertips [100]. However, the speed of this method is limited to the speed of human handwriting, rendering it significantly slower than other text entry methods [91][99]. Handwritten text entry may thus be suitable for brief annotations, but less suitable for longer entries.

Speech recognition software, which employs natural language processing to listen for and interpret speech, is another intuitive method of text entry that may have benefits over typing for short messages [91][101]. However, the viability of the method for CRACCEN is less promising, as the software would need to be able to recognize and distinguish between multiple users in an environment with high ambient noise and highly domain-specific terms and acronyms. One major hurdle for speech recognition software in a collaborative environment is the need for the system to differentiate between verbal inputs intended for the system and ongoing communication between teammates [102].

### 5.1.2 Incorporating 3D information

Because UWW takes place in 3D space, CRACCEN should incorporate some ability to visualize and interact with the AOO in 3D. Section 4 outlined several potential approaches to 3D augmentation. The suitability of 3D displays are discussed briefly below, before turning specifically to VR and AR displays, for which much more research on applications to teamwork exists.

### 5.1.2.1 Stereoscopic and auto-stereoscopic 3D displays

Overall, little work has directly examined the use of stereoscopic and auto-stereoscopic 3D displays by multiple simultaneous users. However, a few conclusions can be inferred from the user experience of such displays for individual users.

Interaction with 3D displays viewed through stereoscopic glasses faces complications from the illusory nature of the 3D image: whereas the image appears in 3D, the surface upon which users can interact with the image remains two-dimensional. This has implications for the accuracy of touch input [103], since a user cannot simultaneously converge her eyes on her finger/stylus at the surface of the screen and on the 3D image of the object, which appears either in front of or behind the screen [104]. The result is that either her finger/stylus or the 3D image will appear blurred and doubled. To resolve this issue, mid-air 3D touch can be supported with cameras that track hand/stylus movement [105][106][107][108][109]. However, the image seen through stereoscopic glasses is distorted and can cause user discomfort when viewed at different vertical or horizontal angles than front and centre [110]. This limitation calls into question the viability of stereoscopic glasses for multiple users interacting around a display, and particularly at the angle required for a tabletop display.

Other technologies such as the Euclideon Holographic Table (Euclideon Pty Ltd) allow multiple users wearing special stereoscopic glasses to view and interact with detailed 3D objects in ways that mimic holography (see also [111][112]). A 3D image appears as projecting from a tabletop surface and cameras track users' glasses in order to display the correct viewing angle as they move around the table. Multiple users are able to see different angles at the same time because the computer projects appropriate viewing angles for all users onto the table at individual ranges of light frequencies. The user's glasses filter the light so that only the frequency range appropriate to their viewing angle is seen. Tracked styluses moreover allow multiple users to interact with the 3D images.

Auto-stereoscopic (AS) systems are limited in the number of "sweet spots" from which the 3D object can be accurately viewed [15]; user numbers are thus limited and viewing angle is restricted. More recent innovations have incorporated eye or head tracking to adjust the display such that the "sweet spot" moves with the user [113], and further innovations in multiple-user tracking have allowed for simultaneous AS 3D views [114][115]. In all cases, these technologies work best when the surface of the screen is approximately parallel to the user's face, and thus place limitations on display orientation and the distribution of multiple users. Because they rely on 2D display surfaces, AS systems generally carry the same restrictions on user interaction as do stereoscopic glasses, although some promise has been shown with motion-tracked mid-air gesture inputs [115]. By contrast, volumetric displays, in which pixels are suspended in midair within an enclosed 3D surface, do not carry the same restrictions on interaction. Users can interact with the 3D images in a limited capacity by touching the external surface of the enclosure [29], and more recent innovations allow a higher level of interactivity with motion-tracked gesture inputs [30] or angle- and pressure-sensitive stylus input [116].

### 5.1.2.2 Extended reality displays

VR HMDs have shown some promise in augmenting SA in collaborative planning tasks, as a complement to traditional planning over 2D maps [117]. Indeed, VR HMD systems are regularly proposed for remote collaboration, as they allow for a shared awareness of virtual environments and facilitate otherwise difficult interactions over a distance (e.g., [118][119]). However, such systems impose limitations on the capacity for co-located collaboration, because the external world, including one's collaborators, is occluded and replaced by a full-view virtual display. Compared to face-to-face collaborative work in a real environment, collaborative work using VR HMDs suffers from an increase in miscommunications and collisions [120]. This is due to an inability to see the actions, gazes, movements, and non-verbal cues of collaborators. The result is a decrease in performance on some collaborative tasks compared to see-through AR HMDs [120]. The effects can be somewhat mitigated by tracking the movement of users' bodies and incorporating a digital avatar for each user into the virtual world [121][122]. But important cues such as gaze direction are not easily read from such avatars and they may not aid task performance [123]. Other approaches to these limitations render a user's pointing (e.g., by finger or input device) as a visible ray in the shared virtual environment, or employ algorithms to combine different users' views for a more complete view of the collaborative virtual environment [124].

VR HMDs may be more appropriate when the majority of the work to be completed is loosely coupled individual work, interspersed with short periods of discussion and tightly coupled collaboration [125]. Some research has proposed combining a single VR HMD with a secondary view of the virtual scene on a handheld AR device, which yields similar communication and performance results as a setup with two VR HMDs [126] (see also [127][128][129]). Research on such asymmetric use of VR HMDs is still in its early stages and may soon provide guidelines on asymmetric setups for different styles of collaborative work [130][131][132].

See-through AR HMDs such as the Microsoft HoloLens2, by contrast, allow for multiple users to safely interact in the same space. Co-located collaborative use of AR HMDs has been demonstrated with upwards of eight simultaneous users [133]. The literature in fact recommends see-through AR HMDs for successful and safe communication between co-located collaborative users [120][134]<sup>7</sup>. Multiple-AR-HMD setups support each:

<sup>&</sup>lt;sup>7</sup> It has been suggested that AR HMDs may nevertheless restrict the expression and perception of nonverbal cues by obstruction of the eyebrows and gaze information [135]; more research on the actual impacts of such obstruction is needed.

1) tightly coupled interactions, wherein each user can manipulate the same 3D scene (e.g., [31][134]); 2) driver-follower interactions in which one user actively controls the scene and other users simply view the changes being made (e.g., [136]); and 3) loosely-coupled interactions, wherein each user manipulates their own independent view (e.g., [137]). This ability to switch between multiple levels of interactivity provides important flexibility for collaborative work. As with VR HMDs, collaboration and communication with AR HMDs can be facilitated by rendering a user's point or gaze as a visible ray in the shared virtual environment [138].

Compared to VR HMDs, users completing tasks using AR HMDs maintain greater awareness of the real world and at levels comparable to completing the same tasks without any HMD [139]. AR HMD systems reduce the separation between task space and communication space, and facilitate natural communication cues to improve both task performance and communication metrics [140][141]. Moreover, the see-through nature of AR HMDS allow for continued interactions and collaboration between users even when some are not wearing HMDs [142]. Such asymmetrical use of AR HMDs in teams may in fact improve communication quality, relative to teams with no AR HMDs [143]. Research has demonstrated an advantage of AR systems, over and above traditional methods of SA and planning, in co-located teams of firefighters, police, and military planning personnel [144]. This use case has obvious parallels with the intended application of CRACCEN. Likewise, the U.S. Navy and Army have recently demonstrated AR HMDs as a useful tool for collaborative command and control tasks [145][146].

AR HMDs may be particularly appropriate in combination with a tabletop display, such that users can move fully around the tabletop to view a 3D scene from every angle without losing sight of the tabletop or their collaborators in the real world (e.g., [147]). This setup would also resolve complications with text orientation on tabletop displays, because AR HMDs can render text in an orientation appropriate to each user. This combination benefits from the rich depth cues provided by AR to visualize information, and the ease and familiarity of touch input on a tabletop display [137]. It has proven a useful combination for urban visualization [148][149], architectural design [150], and aerial mission monitoring [134], all of which involve visualization of 3D objects (i.e., buildings, airplanes) and 3D space extending from the horizontal surface of the ground. UWW could similarly benefit from combining AR HMDs and tabletop displays, as SA in this realm involves visualization of 3D objects (i.e., ships and submarines) and 3D space extending from the surface of the ocean floor.

Handheld AR offers an alternative to HMDs, although interactions with 3D objects via tablets have been shown to be slower, more physically tiring, and less preferable compared to AR HMDs [151]. An examination of handheld AR for a collaborative navigation task found significant advantages of handheld AR maps over traditional 2D virtual maps in facilitating communication, establishing a common understanding, and encouraging discussion [152]. Likewise, allowing for individually manipulable views of a 3D object can aid in teaching and learning [153]. However, handheld AR for group work may be associated with a higher workload, particularly for complex problems, which can in turn hinder communication and collaboration [154]. Indeed, whereas VR HMDs are less effective than fully see-through (e.g., AR) HMDs at promoting effective communication between collaborators [120], both task and communication performance suffer with handheld (e.g., video feed) AR compared to VR HMDs [126]. For a full review of mixed and augmented reality in collaborative task settings, see [155][156].

### 5.1.2.3 Input methods for 3D displays and extended reality

When 3D information is presented on 2D displays, several familiar input methods exist. Direct touch (e.g., on handheld AR devices) and indirect mouse dragging can be used to pan, tilt, and rotate in 3D space, whereas gaming controllers often use dual joysticks to facilitate 3D navigation. Within design industries, the 3D mouse is a common solution for the manipulation of 3D objects on 2D displays, providing six degrees of freedom for movement along all axes (e.g., SpaceMouse by 3DConnexion, Logitech). Gesture-based methods have also been developed for direct touch manipulation of 3D objects on 2D displays [157],

Such input methods are less feasible for interaction with VR and AR HMDs [158]. More commonly, HMDs use either motion-based hand tracking to support virtual "direct" touch in mid-air, or infrared-sensor based tracking

of handheld remotes for the indirect selection and manipulation of virtual objects. Both input methods allow the user to interact with virtual objects in a way that reinforces immersion in the 3D environment.

Most off-the-shelf VR HMDs come equipped with infrared-tracked handheld remotes. Users aim a laser cursor projected from the remote toward a virtual object and can press one of several buttons to perform different interactions. VR HMD systems can also track hand motions for remote-free gesture inputs, and offer comparable performance and usability to handheld remotes [159].

Because of the nature of AR environments, gestures and mid-air "direct" touch are the most common forms of input with AR HMDs. Research has shown that bimanual manipulation of 3D scenes aids users' understanding of the 3D space by providing the user's own body as a natural spatial referent [160]. Although intuitive and natural to perform, pre-defined gestures may not match the user's assumed and preferred function, and may thus be difficult to learn and remember [161][162]. User-defined gestures may circumvent some of these complications [162][163]. The Microsoft HoloLens2 supports accurate onboard recognition of simple hand gestures by the outward facing tracking cameras for input. Other devices, such as the Microsoft Kinect, can be tethered to an AR or VR HMD system in order to expand the range of available gestural inputs and allow for custom user-defined gestures [163].

Tracked remotes are less common with off-the-shelf AR HMDs, but may provide some usability and ergonomic advantages compared to mid-air "direct" touch [164]. Unlike tracked handheld remotes, mid-air "direct" touch suffers from an absence of haptic feedback, which can impair accuracy, speed, and perceived ease of use [164]. More recent work has examined the use of tethered smartphones [165] or tablets [166] as tangible input devices for collaborative work in AR environments. Other forms of feedback, such as auditory cues, could be incorporated to support mid-air manipulation of virtual objects displayed on AR HMDs, but consideration must be made for ambient noise within the shipboard operational environment [167].

Because the particulars of a given 3D input method are tied so closely to the display technology (e.g., proprietary remotes for VR HMDs and built-in gestures for AR HMDs), and because literature searches yielded no results pertaining to suitability of 3D input methods for teamwork and collaboration, the results from this section are not independently considered in the summary table in Annex A.

### 5.1.3 Combining multiple displays

Although CRACCEN's requirements suggest the need for at least one central 2D display, the system need not be limited to a single display technology. Indeed, there may be some benefit to incorporating multiple technologies for complex tasks involving multiple users [168][169]. Research examining collaborative group work considers the concept of coupling, a measure of how involved with one another's work collaborators are (e.g., [170]). For tightly coupled tasks, group members work in close collaboration toward one or several common goals, making decisions and implementing changes by deliberation and consensus at the level of the group. For loosely coupled tasks, group members may work in parallel on different aspects of a problem in order to collectively reach a goal; decisions are made at the individual level or occasionally in looser consultation with other group members. Mixed-focus collaborative tasks involve a mixture of both loose and tight coupling, and involve transitions between the two according to the flow of task requirements [56][84][168]. The workload level of a given task may influence coupling and information-sharing requirements as well. For example, team performance under a high workload may suffer when coupling is either too loose (e.g., no information is being shared between team members), whereas a low workload condition may benefit from looser coupling [171].

The sorts of complex mission planning and execution tasks to be completed by the UWW team using CRACCEN are expected to involve mixed focus and a variety of workload conditions. Because the balance of loose and tight coupling in mixed-focus work might be expected to vary across individual UWW command teams [40], and across tasks and workload conditions, flexibility of the selected technologies may be of general benefit. Combining multiple displays for CRACCEN could facilitate this flexible coupling.

#### DRDC-RDDC-2022-R139

In one possible configuration, handheld devices for individual work could be provided, which can later send data to the central group display for discussion and further collaborative work [34][35][172][173][174][175][176]. Similarly, tethered handheld devices have been demonstrated as alternate input methods to large group displays [175][176][177]. Interestingly, the aforementioned differences between vertical and tabletop orientations in promoting collaboration and communication (e.g., [39]) may be diminished by the introduction of auxiliary handheld devices, which allow work to be more distributed [178]. Docking stands that hold handheld devices at a comfortable height and angle (e.g., [179]) can be placed around the group display to facilitate the aggregation and discussion of individual work. Desktop computers may also be incorporated in a mixed focus collaborative system, and the combination of multiple displays for different aspects of group work has shown promise in complex decision-making tasks [172].

One of the major hurdles for mixed-focus system design is the balance of awareness: ensuring that collaborators remain aware of one another's actions and the group-level state of progress on a task, while also being able to maintain focus on their own individual work [170][180]. Some research has found decreased levels of collaboration when handheld devices were incorporated with tabletops [181]. This deleterious effect may be alleviated by notifications and alerts that help maintain other-awareness [47], while information linkages between devices can help users switch fluidly between devices and task modes [182][183][184][185].

Design for multiple integrating technologies must also consider the more practical challenges of system complexity and technological troubleshooting. Auxiliary devices networked to a central group display and/or to one another may introduce frustration and barriers to use in the form of poor integration, mismatching software updates, bandwidth limitations, network connectivity, device communication issues, and higher maintenance requirements for multiple devices and their integration. The trade-off between system flexibility and complexity must be further examined.

# 5.2 Suitability of display and input technologies for the UWW task environment

The UWW task environment is necessarily constrained by the physical shipboard environment within which UWW command teams will use CRACCEN. Environmental considerations such as ambient noise, physical space, temperature, humidity, and lighting will certainly play a role in how CRACCEN is developed and used. However, CRACCEN is a future technology being developed for future ships, and therefore is not constrained by the physical environment aboard existing ships. This section focuses only on those conditions of the UWW task environment that cannot be resolved by future ship design choices, and which will be useful in differentiating between the different display and input technologies under consideration.

### 5.2.1 Eye strain from display technologies

Though any electronic display can cause eye strain and visual fatigue, especially after prolonged use, evidence suggests that the likelihood and severity of symptoms is greater for AR and VR HMDs than for common 2D displays [186][187][188] (c.f. [189]). Stereoscopic viewing, including via HMDs and stereoscopic glasses systems, may exacerbate eye strain due to a mismatch between vergence (adjusting the rotation of the eyes to converge to an object's distance) and accommodation (adjusting the focus of the eyes to the distance of an object) when viewing virtual 3D objects [190][191][192][193][194].

Although some research suggests that see-through AR HMDs promote better depth estimation, greater interaction precision, and less eye strain than VR HMDS [195][196], the likelihood of eye strain symptoms should be a serious consideration for both AR and VR HMDs. Designers must be aware of best practices to minimize eye strain and improve user experience and wellbeing [197][198]. Common AR and VR HMDs have a set focal distance at either infinity (HTC Vive) or approximately 2 metres (Oculus Rift, Microsoft HoloLens2). Infinite focal distances simplify the mathematics for developers but increase the likelihood of eye strain [199].

Virtual objects that users will need to focus on should be projected at a perceived distance of no less than 0.5 m from users' eyes [200]. Tasks requiring users to switch between more near and more distant interfaces should be avoided, as moving the focus along the z-axis can aggravate vergence-accommodation conflict and strain [199][200][201]. Notably, see-through AR HMD systems may require users to switch focus between the virtual objects projected near the eye and more distant real-world objects. It is recommended that virtual objects in such systems are projected at a perceived distance between 1.2 metres and 5 metres away from the user for optimal comfort and resistance to eye strain [200][202].

Various adaptive algorithms have been developed that may minimize the occurrence of eye strain and visual fatigue from HMD usage by dynamically adjusting the user's view [203][204]. The use of "dark mode" graphics, where light text and graphics are positioned against a dark background, may alleviate eye strain symptoms in VR HMDS [205]. Likewise, employing light-coloured graphics in a dimly lit environment will result in optimal user experience and comfort with AR HMDs [206]. There is also some evidence that training and experience with HMDs can improve eye strain symptoms [207][208].

### 5.2.2 Cybersickness from extended reality HMDs

VR users often experience a phenomenon known as cybersickness or simulator sickness, the symptoms of which include nausea, fatigue, disorientation, headaches, and general discomfort [209][210][211]. The symptoms can last for hours after removing a VR HMD and may ultimately discourage use or hinder performance on tasks in the real world [211][212][213]<sup>8</sup>. Although not fully understood, research suggests that cybersickness is caused by sensory conflict, or mismatches between expectation and perceived reality—whether resulting from insufficient refresh rate of the projected image, latency issues, poor tracking of the user's head motion, or mismatch between the visual and vestibular systems (i.e., the user sees that they are moving quickly through a scene but the vestibular system in the ears does not register any motion) [209][212][215][216][217][218]. Eye movement training immediately prior to VR usage may mitigate the likelihood of cybersickness symptoms [219], although more research is needed to validate this effect with a larger sample size. Likewise, techniques used to condition individuals to motion sickness may alleviate the propensity to experience cybersickness over time [220].

The propensity to develop cybersickness differs between individuals and between VR applications, and much effort has been devoted to developing means of detecting and mitigating cybersickness in real time (e.g., [221][222][223][224]). Because increased length of HMD use is generally associated with increased likelihood and severity of cybersickness symptoms, designers should only consider HMDs as supplementary displays for CRACCEN (c.f. [225]). Multiple review papers have catalogued the high level of individual variability in the experience and time course of cybersickness using HMDs [226][227].

Little research has explored the possible interactions between cybersickness effects in a virtual world and real whole-body motion. Users at sea experience continuous whole-body motion, particularly on rough seas (see Section 5.2.3), which may amplify the mismatch between perception and expectation in a VR environment and thereby increase the risk of cybersickness. Such amplification effects have been observed when passengers in moving vehicles or motion platforms used VR HMDs (c.f. [228]), and efforts to incorporate synchronized visual indicators of the external motion may either alleviate [229][230] or further exacerbate [231] symptoms. Other research has failed to find an increased risk of cybersickness symptoms when AR or VR HMDs were used under conditions of simulated ship motion, compared to no motion. However, the effects of ship motion without an HMD were not assessed so the effects of cybersickness and seasickness cannot be differentiated [232]. As a further complication, some research suggests that VR HMD users may consciously or subconsciously limit their head movements when experiencing shipboard motion in order to offset the felt or anticipated effects of sickness [233]. There is also some evidence that cybersickness from VR HMDs may negatively impact balance and

<sup>&</sup>lt;sup>8</sup> C.f. [214], which showed that impairments to simple and complex reaction times after VR use were only weakly correlated with symptoms of cybersickness, suggesting that VR use was itself responsible, regardless of cybersickness symptoms. This is a concerning finding worthy of further exploration.

postural stability [234][235], which may be a risk factor for accident or injury aboard a moving vessel. Undoubtedly, the interaction cybersickness and ship motion needs further study, particularly within the unique population of experienced naval personnel.

Although AR HMDs do not seem as likely as VR HMDs to induce cybersickness [195][236] (c.f. [237][238], which found no evidence of cybersickness using either VR or AR HMDs), AR HMD users can still experience sickness symptoms caused by eye movements [239]. For example, interpupillary distance varies between individuals and misalignment with the HMD's settings can frustrate stereoscopic accommodation and convergence, leading to symptoms of fatigue and headache in some users [240][241]. Some HMD systems such the HoloLens2 have built-in settings to adjust interpupillary distances for unique users, and can recognize individual users through a locally stored retinal database to automatically adjust interpupillary distance for optimal stereoscopic viewing. Nevertheless, AR HMD use provokes a higher reported incidence of cybersickness than does handheld AR, and some preliminary evidence suggests symptoms could linger after prolonged usage as they do with VR HMDs [239]. More research is needed to validate the health and safety of AR HMD use aboard naval vessels.

### 5.2.3 Impacts of ship motion on input technologies

Turbulent conditions, such as those induced by ship motion on rough seas, can negatively impact the accuracy of input on electronic devices [242][243], whether directly or by indirect means (e.g., via increased fatigue or seasickness [244], or impaired visual tracking [245]). Fine motor movement is particularly affected, which renders direct input (i.e., touch, pen styli) more difficult than indirect methods such as mouse, trackball, or keyboard inputs [246][247]. In studies of simulated ship motion, the accuracy of touch input is shown to dramatically worsen as ship motion increases, particularly for smaller targets and gesture-based inputs [248][249]. Mouse input, by contrast, is fairly resilient to simulated ship motion [248]. Although trackballs are often considered optimal input devices to withstand ship motion, [248] also found that trackball input was consistently slower and less accurate than mouse and touch input, even under heavy simulated ship motion (see also [250]).

In simulations of turbulent aircraft conditions, touchscreen users experience decrements to overall performance, usability, and comfort [251]. This is especially true as the display size decreases, suggesting that a large group display may not suffer from the detrimental effects of motion to the same extent as handheld devices [252]. Touch performance is improved when users can brace their hands or wrists in some way, but this can introduce new ergonomic complications from unnatural or difficult finger reaching [253][254]. It is not known to what extent the accuracy of extended reality gesture inputs would be similarly hindered by ship motion, and future research must explore this possibility. Research has, however, demonstrated significant detrimental impacts on accuracy using physical inputs (i.e., a computerized shooting task) when military members wore either AR or VR HMDs under conditions of simulated ship motion [232]; there was no difference in effect between the two types of HMD.

Of secondary consideration, the use of peripherals for indirect (e.g., mouse) or direct (e.g., pen styli) inputs introduces additional complications from ship movement. Peripherals left unfixed and unattended to could slide away from their display devices and cause damage or loss.

### 5.2.4 Impacts of ship vibration on display and input technologies

Whereas a ship's motion on rough seas can be considered a low-frequency vibration, the sorts of whole-body high-frequency vibrations usually referred to by the term also have an impact on the motor and cognitive abilities of ship personnel and thus must be considered in both display and input technology selection [255]. Higher frequency vibrations such as those produced by engines or other onboard machinery can affect the stability of vision and fine motor skills [256], leading to errors in both perception and input accuracy [257]. A meta-analysis found that higher-frequency whole-body vibrations could have a greater detrimental effect on performance than

lower-frequency vibrations, particularly for tasks requiring a high degree of input accuracy [258], although the amplitude of the vibration may be a more critical factor than its frequency [255]. As with lower frequency ship motion, one would expect that input performance with larger displays, which afford larger targets and thus require less precision to operate, would be less affected by high-frequency vibrations than would smaller displays. Research comparing input methods found that the speed and accuracy of trackball inputs were more negatively affected by engine vibrations than mouse or touch inputs, which did not differ from one another [259].

HMDs could exacerbate the deleterious effects of vibration on vision, since both the eyes and the screen in front of them could experience vibratory jitter. The extent to which the amount of vibration experienced in the operations room would hinder comfortable usage of HMDs is a worthy topic for further study before extended reality HMDs could be incorporated into shipboard systems. Previous use of HMDs in military settings has demonstrated that problems with vibration in airborne and ground vehicles can be ameliorated with careful attention to proper mounting of the HMD to the user's head [260]. This feedback pertained to custom designed US military HMDs, however, and may not be applicable to the systems eventually procured by the RCN.

Finally, if vibratory feedback or alerts are relevant to CRACCEN's functioning, then the ability of users to perceive such feedback through the noise of whole-body vibration needs to be assessed. This is a particularly relevant concern for handheld devices but could also be a factor for soft keyboards and direct touch input on large displays, which may benefit from vibratory keystroke feedback [96][97].

### 5.2.5 Ergonomics of display and input technologies

The ergonomic characteristics of devices are of critical importance for long-term or frequent technology use, as they can greatly impact user morale and wellbeing. It has long been known, for example, that prolonged computer mouse usage is associated with musculoskeletal disorders such as carpal tunnel syndrome (e.g., [261]). Alternative input devices have been developed to alleviate the ergonomic strain, but users tend to prefer the traditional mouse to learning novel input methods, which often require considerable training to show any ergonomic or usability benefits [262][263][264][265]. Trackballs may provide more ergonomic support than traditional mouse input devices, provided that care is taken to reduce extreme wrist postures in the trackball's placement, with consideration for individual user ergonomics [266]. Supports to the forearm and palm may offset the ergonomic strain of standard mouse [267] and keyboard [268] inputs. Comparisons between standard computer mouse and pen stylus inputs (i.e., on a horizontal input pad) with a desktop setup show either no differences in muscular load [269] or greater muscular load from pen stylus use [270], depending on the specifics of the experimental setup. By contrast, pen stylus input on a slanted touch screen tablet, with both forearm and tablet resting on a horizontal surface, induced less wrist strain and higher user comfort than standard computer mouse input [271]. For text input, soft keyboards show an ergonomic advantage relative to physical keyboards in the short term due to decreased typing force, but a disadvantage in the form of increased shoulder strain over the long term [272].

Comparing tablets, laptops, tabletops, vertical displays, and smartphones, only the laptop was found to be suitable for long-term use, and only when a proper seated posture was encouraged [273]. The remaining devices were shown to cause significant strain on shoulders, necks, and arms. The same study showed that large vertical displays, when used from a standing position, were particularly straining and not recommended for long-term use (see also [274]).

Large touchscreen displays have been shown to cause more muscle fatigue when displayed in a vertical rather than tabletop orientation [275][276], which can compound into more lasting damage to the musculoskeletal system. Researchers recommend reducing the extent of continuous interaction required and circumventing the need for distant interactions on large displays by bringing important interfaces closer to the user [275][277]. The reach distance for comfortable touch interactions on a tabletop display, for example, is smaller than the size of many large displays [77], so designers need to consider the layout of interfaces carefully. Touchscreen displays angled at around a 45° tilt cause less self-reported discomfort and greater self-reported usability than either

#### DRDC-RDDC-2022-R139

vertical or horizontal displays [278]. When participants are given a choice of several screen angles they tend to prefer an angle between 30 and 45° [50][279]. Gesturing with both hands rather than using a single hand appears to decrease subjective reports of discomfort when using touchscreens, especially for longer periods of usage [279]. However, more complex gestures, such as pinching to rotate or swiping, are associated with more joint activity than simple tap gestures, which may pose a higher risk of injury [276][280]. The ergonomic constraints of comfortable finger positions need to be considered alongside the display orientation when developing usable gestures and interfaces [281].

The height of a tabletop display strongly influences the ergonomic comfort of its use, with greater vertical distance between the head and the screen leading to increased discomfort and decreased usability [278]; thus, raised screens are preferred when standing and lowered screens are preferred when sitting. Duration of use is a particularly relevant factor for tabletop displays, as the downward angle of the head required for tabletop use can put significant strain on the muscles around the neck and spine [273][282]. Moreover, prolonged reading tasks on a tabletop display can increase perceived visual fatigue and required head movements, thereby adding to neck fatigue, relative to similar tasks performed on a laptop [282]. The method of input likewise plays a role in the ergonomic aspects of tabletop use, with mouse input generally preferred over direct inputs (e.g., touch or pen styli) for long-term use [80], but not necessarily for short-term use [86]. Trackball input may cause decreased discomfort in the hand than mouse input, but the trade-off is increased discomfort in the arms, neck, and shoulders [73]. Handwriting recognition using a pen stylus may alleviate some of the ergonomic concerns surrounding keyboard usage [283], but for the relatively small amount of text input expected to be involved in CRACCEN usage, the ergonomic advantages may make little difference in overall user experience.

It is worth noting that some of the ergonomic complications of tabletop displays may be alleviated by augmentation of the two-dimensional image into 3D. Virtual 3D information can be presented to HMD users at eye-level, rather than requiring them to look down at a tabletop display. Compared to vertical displays, the augmentation of tabletop displays with 3D AR HMDs and carefully designed gestural inputs helps to minimize fatigue by keeping users' hands in a relatively comfortable position between the shoulder and the waist [284][285]. However, mid-air interaction may be prone to cause arm fatigue [160][286] (c.f. [287], which showed no additional fatigue when users played a 2D game with mid-air gestures compared to a standard mouse). Other research has found that gestures performed in a virtual world as seen through VR HMDs resulted in higher levels of self-reported fatigue than the same gestures performed in the real world [288]. Methods for input amplification can reduce the strain of muscles from mid-air inputs [289].

VR and AR HMDs can also put additional strain on head and neck muscles [290][291][292][293][294] and cause discomfort and fatigue [294][295][296], though the amount of user-reported discomfort seems to depend heavily on the overall weight and weight distribution of the HMD [293][297][298]. In general, prolonged use is not recommended. A recent review contracted by DRDC suggests that AR HMDs such as the Microsoft Hololens2 should be generally safe for use in UWW, though ergonomic considerations suggest the duration of use be limited [297].

By comparison, 3D AR interactions are more physically tiring when performed via handheld devices rather than via HMDs [151]. Research has found significant neck, back, and arm strain following prolonged use of handheld devices [61][273][299]. Discomfort in the neck and back is especially pronounced when handheld devices are positioned in the lap while sitting [300], whereas holding a device aloft in one hand and performing touch actions with the other puts significant strain on the supporting wrist [301]. The ergonomic disadvantages of handheld devices may be alleviated by incorporating docking stands that support the device and allow for two-handed touch input at a comfortable height and angle relative to the user [302], but handheld AR devices generally need to be held aloft and moved to view a 3D scene.

# 6 **Recommendations**

The requirements analysis for CRACCEN outlined in Section 2 identified the need for at least one central 2D group display and the ability to visualize and interact with 3D information, given the 3D nature of the UWW environment. The possibility of combining multiple displays to accommodate the flexible transition between individual and group aspects of UWW task work was also discussed. Sections 4 and 5 presented a broad and interdisciplinary review of literature, in an effort to determine which display and input technologies are best suited to facilitate teamwork in the UWW environment.

A table summarizing the findings of the review is presented in Annex A, which facilitates direct comparisons between technologies. A full list of the references that guided the assessment of each technology is provided in the summary table and, for conciseness, is not reproduced in the written recommendations below. The recommendations drawn from the review are discussed in turn as follows:

- 1. A large tabletop 2D display should serve as the central group display;
- 2. Mounting hardware should be selected to allow the central display to convert flexibly between horizontal and vertical orientations, as needed; and
- 3. Optional additional devices could be networked with the central display to improve 3D visualization of the battlespace (i.e., AR HMDs) and to better support individual work that can subsequently be combined and shared on the central display (i.e., handheld devices or desktop workstations).

# 6.1 Large central tabletop 2D display

Working around a table is a familiar configuration for group discussion, deliberation, information sharing, and decision-making, and the intuitiveness of a multi-touch screen makes it easy for all users to contribute, whether simultaneously or in turn. This layout enhances communication and interaction quality in collaborative tasks and the ability to interact with the display from any edge greatly increases the simultaneous user capacity. The technology shares similarities with familiar and regularly used technologies such as desktop computers and smartphones, which will allow users to leverage existing knowledge frameworks and thereby reduce learning time. Knowledge frameworks for map reading, which often occurs on a horizontal surface, can also be leveraged for spatio-navigational planning and SA.

Direct touch is the ideal input method for 2D tabletop displays. This is particularly true when group work is required, because it allows users to easily follow the actions of their co-collaborators. Precision can be increased with pen styli, which can moreover be used for handwriting recognition as a text input method, though consideration for ship motion and moving peripherals may outweigh any added value from styli. Soft keyboards can be implemented for use by UWW teams as needed.

Special consideration for the ergonomics of use must be made during design phases, ensuring that important display menus are easily accessible and/or can be flexibly moved around the screen for easy access. Tabletop 2D displays have the potential to place strain on the neck and spine if positioned too low relative to the user, and touch-based interactions on large displays can cause muscle fatigue with extended use.

# 6.2 Flexible display angle

With the need to support mixed-focus tasks, flexibility would greatly benefit CRACCEN's usability. At a minimal extra cost for specialized mounting hardware, a collaboration-focused tabletop display can be converted into a presentation-focused vertical display on an as-needed basis. The vertical orientation would support more lecture-style communications; would facilitate briefings; and could stand as a background display to maintain

#### DRDC-RDDC-2022-R139

SA of tasks, platforms, and mission status when not actively being used for collaborative mission planning or execution tasks.

Special consideration for flexible interfaces must be made during design phases, as the orientation of the screen will greatly impact the type of information to be displayed and the organization of the interaction space. For example, reaching a menu in the top-left corner may be a trivial task for a user on the top edge of a tabletop display but a very taxing one for a user situated in front of a vertical display. The flexible system should be able to automatically recognize the orientation of the display and adjust output and interfaces accordingly. Although existing research provides some general guidelines (e.g., [38][39]), future research exploring specific naval use cases for different display orientations should guide CRACCEN interface design.

# 6.3 Optional AR HMDs and devices for individual work

Because the UWW environment is 3D, SA and mission planning are likely to benefit greatly from its 3D representation. Although both VR and AR HMDs support immersive interaction with virtual 3D objects, see-through AR HMDs provide better supports to communication and group work as well as a lower likelihood of cybersickness. AR HMDs could be available to CRACCEN users as optional aids for 3D visualization and interaction, and research suggests that effective collaboration and communication can be maintained at a high level even in instances where some collaborators wear HMDs and others do not.

Use cases for AR HMDs have already been explored in a variety of military settings, including navigation and SA in urban environments [303][304], army ground and vehicle warfighting scenarios [260], heads-up displays for air force pilots [260], and simulated training for high-risk scenarios [305]. To date little work has assessed the usability of AR HMDs in shipboard settings (but see [232][306]). Before development proceeds, research must be conducted to test the feasibility of AR HMDs in shipboard environments, as the effects of ship motion and vibration may hinder or preclude usability from the outset.

Although the focus of CRACCEN is on group work, individual UWW team members will still need to apply their individual expertise to the collaborative mission planning and decision-making tasks. Tablets or desktop computer systems could be available as optional supports to individual work, with networking capabilities to transmit information to the central group display during active mission planning tasks. The scientific literature does not currently provide clear guidelines for the suitability of one device (e.g., desktop or tablet) over the other for mixed-focus tasks, so both should be considered in initial research. Previous work combining technologies such as tabletops, tablets, 3D augmentation, and AR HMDs, for multi-user visualization in archaeology serves as a useful proof of concept: providing multiple methods for exploring and navigating a complex 3D environment can aid visualization and SA and thereby support interpretation and decision-making [307].

Designers of a flexibly coupled CRACCEN system will need to consider the best interfaces for individual work, how to transmit information between displays, and how transmitted information will be incorporated into the visualizations on the central group display. The trade-off between system flexibility and the additional practical complexity of integrating multiple devices (e.g., inter-device communication challenges, software updates, bandwidth limitations) must also be carefully considered.

# 6.4 Future research

Additional research is required to validate the above recommendations within the unique operational setting of shipboard UWW command teams of the RCN. Future research proposals for next steps in the development process are presented below in order of priority, although several lines of research may be conducted concurrently. In all proposed research cases, prototype development and user research are to occur in tandem, with each informing the other in an ongoing and iterative fashion. Continuous development and refinement of prototypes is assumed at all stages of the future research proposed.

### 6.4.1 Ideation studies for flexible displays

Thus far, this report has recommended the use of a central 2D tabletop display, the option to orient it vertically, optional AR HMDs for 3D information, and optional desktop or tablet devices for individual work. However, it is not clear which information and capabilities would best be presented on which display(s) and in which combination(s). An initial ideation study could have SMEs perform a brainstorming activity wherein they imagine conducting various mission planning, execution, and decision-making tasks on the different display options. Questions of interest in this line of research include: which tasks SMEs would like to be able to complete with each display option, which display option(s) they might look to when seeking particular information, what problems or issues they foresee with a given display, and what contribution they can imagine for their own role within the UWW team using each of the display options.

Building on the results of this initial ideation activity, follow-on research should assess design concepts and static mock-ups for interfaces on each display. Initial mock-ups and concepts would be developed from existing UX expertise, and subsequently presented for user testing and refinement. Questions of interest at this stage include: the perceived usability of the interface layouts presented, what changes SMEs might recommend to improve the layout of a given display, what changes SMEs might recommend with regard to the distribution of information or capabilities across displays (e.g., validating the initial ideation study), and how SMEs imagine themselves interacting with the displays. It is likely that several iterations of design concept development and user testing will be required, and it is possible that these initial studies will conclude counter to the recommendations provided in this preliminary report. It may be the case, for example, that existing desktop workstations available to the UWW team might prove sufficient for the capability requirements of the individual work aspects of CRACCEN, in which case further development of individual desktop or tablet interfaces need not proceed. In this way, the results of these studies will inform and constrain the direction of the additional research recommendations below.

Researchers might also consider employing VR to test design concepts with SMEs in a simulated shipboard environment. This would allow SMEs to experience and assess, at an early stage of development, the look and feel of CRACCEN within a virtual mock-up of their real work environment, rather than in a less realistic laboratory setting [308][309].

### 6.4.2 Testing prototypes for the central display

The next step recommended by the findings of this report is to develop a low-fidelity initial prototype for a CRACCEN tabletop, focusing on a usable 2D interface with a mock-up scenario that UWW team can work through as a group. Researchers can then use this prototype to test the usability of the system with SMEs and teams of SMEs [310].

Early iterations of prototype design and development may focus on high-level goals, and early testing may likewise involve relatively simple feedback methods, such as the Feature Capture Grid, which organizes feedback into likes, criticisms, questions, and ideas, in order to refine later design iterations [6].

Subsequent prototype iterations should incorporate increasingly detailed goals and specific capabilities. User testing of these later iterations could incorporate measures of task performance, usability (e.g., Tabletop Collaborative Usability Assessment [311]), verbal walk-throughs of system usage (e.g., [312]), post-scenario interviews, measures of workload and cognitive load (e.g., NASA-TLX [313]), and measures of team communication and cohesion. Eye-tracking measures may be incorporated to help improve interface design [314]. After a period of familiarization with the prototype, as recommended by [315], a formal evaluation of Task Technology Fit (TTF) should be undertaken to validate its fit to the task domain [316][317][318]. If the results of the ideation studies described above validate the usefulness of an optional vertical orientation, testing of this prototype should examine designs for this orientation as well.

### 6.4.3 Testing AR HMD and individual work prototypes

The feasibility of AR HMDs in shipboard environments must be tested at sea. Provided these tests demonstrate that AR HMDs can be safely used in the settings for which they are intended, development and testing of prototypes should proceed in much the same way as recommended for the central display: moving from high-level goals and simple feedback to tests of specific capabilities with a full suite of measures. Development should consider design heuristics in order to minimize ergonomic strain and maximize usability [319]. Initial evaluations may examine the AR HMD in isolation, but later research must incorporate the prototype of the tabletop display as well. SMEs can be individually tasked with providing initial high-level assessments and input to the use of AR HMDs for CRACCEN.

For collaborative tasks, the level of virtuality and immersion of a mixed reality system ought to be increased only as required by task demands, in order for users to maintain awareness of the physical space and collaborators [17]. The cycle of testing and development for AR HMDs needs to examine not only usability and TTF, but also a comparison of the value added by incorporating HMDs and 3D information [320], relative to any changes to the quality of task performance, cognitive load, teamwork, or communication.

If the results of the ideation studies described above validate the usefulness of tablets or desktop computers for individual work, testing of this prototype should likewise proceed from high-level goals and simple feedback to tests of specific capabilities with a full suite of measures. Initial evaluations may examine the individual work prototype in isolation, but later research must incorporate the prototype of the tabletop display as well. Because these devices are intended for individual work, initial evaluations can be conducted with individual SMEs.

Testing and development for individual work prototypes needs to examine usability, TTF, and a comparison of the value added by incorporating these additional devices, relative to any changes to the quality of task performance, cognitive load, teamwork, or communication.

# 7 Conclusions

The CRACCEN system is expected to revolutionize the way UWW is conducted within the RCN, with the potential to reshape not only how technology is used for UWW tasks but also how relevant tasks are distributed and completed among command team members. What the system should look like in order to accommodate this reimagined task and personnel space is a critical question, as the decisions made in early high-level design phases could constrain and define the nature of UWW in the future of the RCN. This report sought to identify the display and input technologies best suited for use by shipboard UWW teams by reviewing these technologies in relation to a number of high-level task requirements and constraints.

Initial requirements analysis presented in Section 2.3 suggested the need for at least one central 2D display screen, the ability to display and interact with 3D information, and the possibility of incorporating multiple displays for flexible mixed-focus work. This report reviewed multiple display technologies, input methods, and approaches to 3D augmentation, considering the suitability of the various technologies to both teamwork and the constraints of the UWW task environment.

Based on the results of an integrative literature review, the author recommends that a 2D tabletop display be adopted as the central display screen for CRACCEN, with flexible display orientations, optional AR HMDs to augment visualizations into the three dimensions of the UWW environment, and optional desktop or handheld displays for individual work that can be subsequently shared for planning and discussion at the central team display. However, characteristics unique to the UWW command team and their operational environment may influence the utility and usability of these technologies and technology combinations. Further research with SMEs must validate these technologies and test the usability of interface designs as prototype development proceeds.

# References

- [1] CRACCEN (Amendment 1), 01cz (UWW Directed Client Support), signed 13 December 2016.
- [2] Tollefsen, C., Hunter, A., Randall, T., & Isenor, A. W. (2017). Recommendations for implementation of the Command Reconnaissance Area Coordination and Control Environmental Network (CRACCEN) project. Defence Research and Development Canada, Reference Document, DRDC-RDDC-2017-D101.
- [3] Kramer, C., & Baker, K. (2017). Maritime information warfare: Ship command team task characterization for planning. Defence Research and Development Canada, Contract Report, DRDC-RDCC-2017-C116.
- [4] Department of National Defence (2014). NAVORD 3771-12. Defense Research and Development Canada (DRDC) Directed Client Support; Appendix A to Annex A.
- [5] Endsley, M. R., Bolté, B., & Jones, D G. (2003). Designing for situation awareness: An approach to user-centered design. London, UK: Taylor & Francis.
- [6] Randall, T., Hunter, A., & Coady, D. (2018). Underwater warfare mission planning and execution challenges and technology applications: A first look. Defence Research and Development Canada, Reference Document, DRDC-RDDC-2018-D125.
- [7] Lawrynczyk, A., & Baker, K. (2017). Maritime information warfare commander task group future command and control system requirements. Defence Research and Development Canada, Contract Report, DRDC-RDDC-2017-C183.
- [8] Benaskeur, A., Irandoust, H., Baker, K., & Banbury, S. (2009). Naval force-level tactical command & control: Goal hierarchy and analysis. DRDC Valcartier, Technical Report, DRDC-VALCARTIER-TR-2009-197.
- [9] Coates, C., & Kobierski, B. (2006). Human factors analyses of operator positions in the operations room of the HALIFAX Class Frigate. DRDC Toronto, Contract Report, DRDC-TORONTO-CR-2006-117.
- [10] Baker, K., Banbury, S., & Scipione, A. (in press). Command Reconnaissance Area Control and Coordination Environmental Network (CRACCEN) requirements analysis summary report. Defence Research and Development Canada, Contract Report, C22-0420-06975.
- [11] Banbury, S., & Baker, K. (2019). Underwater Warfare (UWW) personnel profiles and system ideation. Defence Research and Development Canada, Contract Report, DRDC-RDDC-2019-C201.
- [12] Torraco, R. J. (2005). Writing integrative literature reviews: Guidelines and examples. *Human Resource Development Review, 4*, 356–367. doi:10.1177/1534484305278283.
- [13] Haddaway, N. R., Collins, A. M., Coughlin, D., & Kirk, S. (2015). The role of Google Scholar in evidence reviews and its applicability to grey literature searching. *PLOS ONE*, 10, article e0138327. doi:10.1371/journal.pone.0138237.
- [14] Grossman, T., & Wigdor, D. (2007). Going deeper: A taxonomy of 3D on the tabletop. *Proceedings of the* 2<sup>nd</sup> *IEEE International Workshop on Horizontal Interactive Human Computer Systems*, 137–144, Newport, RI: IEEE Computer Society.

- [15] Holliman, N. S., Dodgson, N. A., Favalora, G. E., & Pockett, L. (2011). Three-dimensional displays: A review and application analysis. *IEEE Transactions on Broadcasting*, 57, 362–371. doi:10.1109/TBC.2011.2130930.
- [16] Coltekin, A., Lochhead, I., Madden, M., Christophe, S., Devaux, A., Pettit, C.,... & Hedley, N. (2020). Extended reality in spatial sciences: A review of research challenges and future directions. *ISPRS International Journal of Geo-Information*, 9, article 439. doi:10.3390/ijgi9070439.
- [17] Roo, J. S., & Hachet, M. (2017). One reality: Augmenting how the physical world is experienced by combining multiple mixed reality modalities. *Proceedings of the 30<sup>th</sup> ACM Symposium on User Interface Software and Technology (UIST 2017)*, 787–795, Quebec City, QC: Association for Computing Machinery. doi:10.1145/3126594.3126638.
- [18] Dan, A., & Reiner, M. (2017). EEG-based cognitive load of processing events in 3D virtual worlds is lower than processing events in 2D displays. *International Journal of Psychophysiology*, 122, 75–84. doi:10.1016/j.ijpsycho.2016.08.013.
- [19] Jorke, H., & Fritz, M. (2005). Infitec: A new stereoscopic visualization tool by wavelength multiplex imaging. *Journal of Three Dimensional Images*, 19, 50–56.
- [20] Fergason, J. (2008). "Monitor for showing high-resolution and three dimensional images and method." US Patent 7401923.
- [21] Sharp, G. D., & Robinson, M. G. (2007). Enabling stereoscopic 3D technology. Proceedings of SPIE Stereoscopic Displays and Virtual Reality Systems XIV at Electronic Imaging 2007, vol 6490, article 69400X. doi:10.1117/12.705262.
- [22] Nelson, J., & Brott, R. L. (2010). "Autostereoscopic display with fesnel lens element and double sided prism film adjacent a backlight having a light transmission surface with left and right eye sources at opposing ends modulated at a rate of at least 90 Hz". US Patent 7750982.
- [23] Harold, J., Wilkes, D., & Woodgate, G. (2004). Switchable 2D/3D display: Solid phase liquid crystal microlens array. *Proceedings of the 2004 International Display Workshops*, 11, 1495–1946.
- [24] Cobb, J. (2005). Autostereoscopic desktop display: An evolution of technology. Proceedings of SPIE Stereoscopic Displays and Virtual Reality Systems XII at Electronic Imaging 2005, vol 5664. doi: <u>https://doi.org/10.1117/12.585053</u>.
- [25] Speranza, F., Tann, W. J., Martin, T., Stelmach, L., & Ahn, C. (2005). Perceived smothness of viewpoint transition in multi-viewpoint stereoscopic displays. *Proceedings of SPIE Stereoscopic Displays and Virtual Reality Systems XII at Electronic Imaging 2005*, vol 5664. doi:10.1117/12.587170.
- [26] Schmidt, A., & Grasnick, A. (2002). Multiviewpoint autostereoscopic displays from 4d-Vision GmbH. Proceedings of SPIE Stereoscopic Displays and Virtual Reality Systems IX, vol 4660.
- [27] Matusik, W., & Pfister, H. (2004). 3D TV: A scalable system for real-time acquisition, transmission, and autostereoscopic display of dynamic scenes. ACM Transactions on Graphics, 23, 814–824. doi:10.1145/1015706.1015805.
- [28] Park, J.-H., Hong, K., & Lee, B. (2009). Recent progress in three-dimensional information processing based on integral imaging. *Applied Optics*, 48, H77–H94. doi:10.1364/AO.48.000H77.

- [29] Favalora, G. E. (2005). Volumetric 3D displays and application infrastructure. *IEEE Computer*, *38*, 37–44. doi:10.1109/MC.2005.276.
- [30] Grossman, T., Wigdor, D., & Balakrishnan, R. (2004). Multi-finger gestural interaction with 3D volumetric displays. UIST 04: Proceedings of the 17th Annual ACM symposium on user interface software and technology, 61–70. doi:10.1145/1029632.1029644.
- [31] Garneau, C. J., Boyce, M. W., Shorter, P. K., Vey, N. K., & Amburn, C. R. (2018). The Augmented REality Sandtable (ARES) research strategy. US Army Research Laboratory: Human Research & Engineering Directorate, ARL-TN-0875.
- [32] Mateescu, M., Pimmer, C., Zahn, C., Klinkhammer, D., & Reiterer, H. (2021). Collaboration on large interactive displays: A systematic review. *Human-Computer Interaction*, 36, 243–277. doi:10.1080/07370024.2019.1697697.
- [33] Inkpen, K., Hawkey, K., Kellar, M., Mandryk, R., Parker, K., Reilly, D., et al. (2005). Exploring display factors that influence co-located collaboration: Angle, size, number, and user arrangement. *Proceedings of the 11<sup>th</sup> International Conference on Human-Computer Interaction*, Las Vegas, NV.
- [34] Zagermann, J., Pfeil, U., Radle, R., Jetter, H.-C., Klokmose, C., & Reiterer, H. (2016). When tablets meet tabletops: The effect of tabletop size on around-the-table collaboration with personal tablets. *Proceedings* of the SIGCHI Conference on Human Factors in Computing Systems (CHI 2016), 5470-5481, San Jose, CA: Association for Computing Machinery. doi:10.1145/2858036.2858224.
- [35] Wallace, J. R., Scott, S. D., & MacGregor, C. G. (2013). Collaborative sensemaking on a digital tabletop and personal tablets: Prioritization, comparisons, and tableaux. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, 3345–3354, Paris, France: Association for Computing Machinery. doi:10.1145/2470654.2466458.
- [36] Dudfield, H. J., Macklin, C., Fearnley, R., Simpson, A., & Hal, P. (2001). Big is better? Human factors issues of large screen displays with military command teams. 2001 People in Control: 2<sup>nd</sup> International Conference on Human Interfaces in Control Rooms, Cockpits, and Command Centres, 304–309. doi:10.1049/cp:20010480.
- [37] Gouin, D., Vernik, R., & Wark, S. (2011). Using large group displays to support intensive team activities in C2. DRDC Valcartier, Scientific Letter, DRDC-VALCARTIER-SL-2011-338.
- [38] Pavylovich, A., & Stuerzlinger, W. (2008). Effect of screen configuration and interaction devices in shared display groupware. *Proceedings of the 3<sup>rd</sup> ACM International Workshop on Human-centered Computing* (HCC '08), 49-56, Vancouver, BC. doi:10.1145/1462027.1462035.
- [39] Rogers, Y., & Lindley, S. (2004). Collaborating around vertical and horizontal large interactive displays: Which way is best? *Interacting with Computers*, *16*(6), 1133–1152. doi:10.1016/j.intcom.2004.07.008.
- [40] Isenberg, P., Fisher, D., Paul, S., Morris, M. R., Inkpen, K., & Czerwinski, M. (2012). Co-located collaborative visual analytics around a tabletop display. *IEEE Transactions on Visualization and Computer Graphics*, 18(5), 689-702. doi:10.1109/TVCG.2011.287.
- [41] Schubert, M., Serna, A., & George, S. (2012). Using collaborative activities on tabletops to enhance learning and knowledge transfer. *IEEE 12<sup>th</sup> International Conference on Advanced Learning Technologies*, 610–612. doi:10.1109/ICALT.2012.73.

- [42] Buisine, S., Besacier, G., Aoussat, A., & Vernier, F. (2012). How do interactive tabletop systems influence collaboration? *Computers in Human Behavior*, 28, 49–59. doi:10.1016/j.chb.2011.08.010.
- [43] Zaphiris, P., Loannou, A., Loizides, F., & Vasiliou, C. (2013). User experience in using surface computing for collaborative decision making. *Interactive Technology and Smart Education*, 10, 297–308. doi:10.1108/ITSE-11-2012-0030.
- [44] Reski, N., Nordmark, S., & Milrad, M. (2014). Exploring new interaction mechanisms to support information sharing and collaboration using large multi-touch displays in the context of digital storytelling. *IEEE 14<sup>th</sup> International Conference on Advanced Learning Technologies*, 176–180. doi:978-1-4799-4038-7.
- [45] Scott, S. D. (2003). Territory-based interaction techniques for tabletop collaboration. In *Conference Companion of the ACM Symposium on User Interface Software and Technology (UIST'03)*. New York, NY: Association for Computing Machinery.
- [46] Scott, S. D., Carpendale, M. S. T., & Inkpen, K. (2004). Territoriality in collaborative tabletop workspaces. Proceedings of the 2004 ACM conference on computer supported cooperative work (CSCW '04). Association for Computing Machinery, New York, NY, 294–303. doi:10.1145/1031607.1031655.
- [47] Niu, S., McCrickard, D. S., & Harrison, S. (2017). Investigating notifications and awareness for multi-user multi-touch tabletop displays. In Bernhaupt, R. et al. (Eds.), *Human-Computer Interaction (INTERACT 2017). Lecture Notes in Computer Science, 10515*, 223–244. Chamonix, Switzerland: Springer.
- [48] Clayphan, A., Martinez-Maldonado, R., Tomitsch, M., Atkinson, S., & Kay, J. (2016). An in-the-wild study of learning to brainstorm: Comparing cards, tabletops, and wall displays in the classroom. *Interacting with Computers, 28*, 788–810. doi:10.1093/iwc/iww001.
- [49] Jakobsen, M. R., & Hornbaek, K. (2014). Up close and personal: Collaborative work on a high-resolution multitouch wall display. ACM Transactions on Computer-Human Interaction, 21, Article 11. doi:10.1145/2576099.
- [50] Müller-Tomfelde, C., Wessels, A., & Schremmer, C. (2008). Tilted tabletops: In between horizontal and vertical workspaces. *Proceedings of the 3<sup>rd</sup> IEEE International Workshop on Horizontal Interactive Human Computer Systems (TABLETOP '08)*, 49–56, Amsterdam, Netherlands: IEEE Computer Society. doi:10.1109/TABLETOP.2008.4660183.
- [51] Kruger, R., Carpendale, S., Scott, S. D., & Greenberg, S. (2003). How people use orientation on tables: Comprehension, coordination, and communications. *Proceedings of the 2003 International ACM SIGGROUP Conference on Supporting Group Work*, 369–378, Sanibel Island, FL: Association for Computing Machinery. doi:10.1145/958160.958219.
- [52] Mi, M., Tano, S., Hashiyama, T., Ichino, J., Iwata, M., Misawa, J., ... & Yonemoto, K. (2013). A pre-experiment on effects of horizontal and vertical touch displays on group work in card classification tasks. *IEEE 4<sup>th</sup> International Conference on Cognitive Infocommunications (CogInfoCom 2013)*, 173–178. doi:10.1109/CogInfoCom.2013.6719235.
- [53] Barnkow, L., & von Luck, K. (2012). Semiautomatic and user-centered orientation of digital artifacts on multi-touch tabletops. In *Proceedings of the 11th International Conference on Enetertainment Computing* (ICEC 2012). Berlin: Springer-Verlag, 381–388. doi:10.1007/978-3-642-33542-6\_34.

- [54] Hancock, M. S., Vernier, F. D., Wigdor, D., Carpendale, S., & Shen, C. (2006). Rotation and translation mechanisms for tabletop interaction. *Proceedings of the 1<sup>st</sup> IEEE International Workshop on Horizontal Interactive Human Computer Systems (TABLETOP '06)*, 79–88, Adelaide, Australia: IEEE Computer Society. doi:10.1109/TABLETOP.2006.26.
- [55] Schlatter, O., Migge, B., & Kunz, A. (2012). User-aware content orientation on interactive tabletop surfaces. In *Proceedings of the 2012 International Conference on Cyberworlds (CW '12)*. IEEE Computer Society, 246–250. doi:10.1109/CW.2012.43.
- [56] Tang, A., Tory, M., Po, B., Neumann, P., & Carpendale, S. (2006). Collaborative coupling over tabletop displays. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '06)*, 1181–1190, Montreal, QC: Association for Computing Machinery. doi:10.1145/1124772.1124950.
- [57] Hare, C., Sharples, S., Stedon, A., & Talbot-Jones, P. (2013). Four go mad in the Yorkshire dales: Investigating interactions on tabletop displays. In Anderson, M. (Ed.) Contemporary Ergonomics and Human Factors 2013: Proceedings of the International Conference on Ergonomics & Human Factors 2013, 27–34. Taylor & Francis. doi:10.1201/b13826-12.
- [58] Szymaski, R., Goldin, M., Palmer, N., Beckinger, R., Gilday, J., & Chase, T. (2008). Command and control in a multitouch environment. In *Proceedings of the 26<sup>th</sup> Army Science Conference*.
- [59] Forlines, C., Wigdor, D., Shen, C., & Balakrishnan, R. (2007). Direct-touch vs. mouse input for tabletop displays. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI 2007)*, 647–656, San Jose, CA: ACM. doi:10.1145/1240624.1240726.
- [60] Sasangohar, F., MacKenzie, I. S., & Scott, S. D. (2009). Evaluation of mouse and touch input for a tabletop display using Fitt's reciprocal tapping task. *Proceedings of the Human Factors & Ergonomics Society Annual Meeting*, 53, 839–843. doi:10.1177/154193120905301216.
- [61] Toy, K. J., Peres, S. C., David, T. Y., Nery, A., & Phillips, R. G. Jr. (2012). Examining user preferences in interacting with touchscreen devices. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 56, 1862–1866. doi:10.1177/1071181312561375.
- [62] Cockburn, A., Ahlstrom, D., & Gutwin, C. (2012). Understanding performance in touch selections: Tap, drag, and radial pointing drag with finger, stylus, and mouse. *International Journal of Human-Computer Studies*, 70, 218–233. doi:10.1016/j.ijhcs.2011.11.002.
- [63] Sears, A., & Shneiderman, B. (1991). High precision touchscreens: Design strategies and comparisons with a mouse. *International Journal of Man-Machine Studies*, 34(4), 593–613. doi:10.1016/0020-7373(91)90037-8.
- [64] Besançon, L., Issartel, P., Ammi, M., & Isenberg, T. (2016). Usability comparison of mouse, touch, and tangible inputs for 3D data manipulation. (Cornell University Technical Report 1603.08735, arXiv.org). <u>https://arxiv.org/pdf/1603.08735.pdf</u> (accessed January 16, 2020).
- [65] Kin, K., Agrewala, M., & DeRose, T. (2009). Determining the benefits of direct-touch, bimanual, and multifinger input on a multitouch workstation. *Proceedings of Graphics Interface (GI '09)*, 119–124, Kewlowna, BC: Canadian Information Processing Society. doi:10.5555/1555880.1555910.

- [66] Wigdor, D., Penn, G., Ryall, K., Esenther, A., & Shen, C. (2007). Living with a tabletop: Analysis and observations of long term office use of a multi-touch table. *Proceedings of the 2<sup>nd</sup> IEEE International Workshop on Horizontal Interactive Human-Computer Systems (TABLETOP '07)*, 60–67, Newport, RI: IEEE Computer Society. doi:10.1109/TABLETOP.2007.33.
- [67] Noah, B., Li, J., & Rothrock, L. (2017). An evaluation of touchscreen versus keyboard/mouse interaction for large screen process control displays. *Applied Ergonomics*, 64, 1–13. doi:10.1016/j.apergo.2017.04.015.
- [68] Roudaut, A., Huot, S., & Lecolinet, E. (2008). TapTap and MagStick: Improving one-handed target acquisition on small touch-screens. *Proceedings of the Working Conference on Advanced Visual Interfaces* (AVI '08), 146–153, Naples, Italy: Association for Computing Machinery. doi:10.1145/1385569.1385594.
- [69] Buschek, D., Kinshofer, J., & Alt, F. (2017). A comparative evaluation of spatial targeting behaviour patterns for finger and stylus tapping on mobile touchscreen devices. *Proceedings of the ACM on Interactive, Mobile, Wearable, and Ubiquitous Technologies, 1*, Article 126. doi:10.1145/3161160.
- [70] Gupta, A. M., & Bischof, W. F. (2014). Exploring and understanding unintended touch during direct pen interaction. *ACM Transactions on Computer-Human Interaction*, *21*, 1–39. doi:10.1145/2674915.
- [71] Niiro, Y., Kallman, M., & Arif, A. S. (2019). An experimental comparison of touch and pen gestures on a vertical display. In *Proceedings of the 8th ACM International Symposium on Pervasive Displays (PerDis '19)*. Association for Computing Machinery, New York, NY, USA, Article 18, 1–6. doi:10.1145/3321335.3324936.
- [72] Annett, M., & Bischof, W. F. (2015). Hands, hover, and nibs: Understanding stylus accuracy on tablets. In Proceedings of the 41<sup>st</sup> Graphics Interface Conference (GI '15). Canadian Information Processing Society, 203–210.
- [73] Heber, I. A., Oehl, M., & Sutter, C. (2011). Performance and comfort when using motion-controlled tools in complex tasks. In Stehanidis, C. (Ed.) HCI International 2011—Posters' Extended Abstracts. HCI 2011. Communications in Computer and Information Science, vol 174. Berlin, Germany: Springer.
- [74] Robertson, G., Czerwinski, M., Baudisch, P., Meyers, B., Robbins, D., Smith, G., & Tan, D. (2005). The large-display user experience. IEEE Computer Graphics and Applications, 25(4), 44–51. doi:10.1109/MCG.2005.88.
- [75] Slocum, J. (2005). A breakdown of the psychomotor components of input device usage. Usability News, 7(1): 1–6. <u>http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.531.5525&rep=rep1&type=pdf</u> (accessed January 16, 2020).
- [76] Chang, H., Tsai, T., Chang, Y., & Chang, Y. (2014). Touch panel usability of elderly and children. Computers in Human Behavior, 37, 258–269. doi:10.1016/j.chb.2014.04.050.
- [77] Toney, A., & Thomas, B. H. (2006). Considering reach in tangible and table top design. Proceedings of the 1<sup>st</sup> IEEE International Workshop on Horizontal Interactive Human Computer Systems (TABLETOP '06), 57–58, Adelaide, Australia: IEEE Computer Society. doi:10.1109/TABLETOP.2006.9.
- [78] Schick, A., van de Camp, F., Ijsselmuiden, J., & Stiefelhagen, R. (2009). Extending touch: Towards interaction with large-scale surfaces. *Proceedings of the ACM International Conference on Interactive Tabletops and Surfaces (ITS '09)*, 117–124. Banff, AB: Association for Computing Machinery. doi:10.1145/1731903.1731927.
- [79] Suto, S., Watanabe, T., Shibusawa, S., & Kamada, M. (2018). Multi-touch tabletop system using infrared image recognition for user position identification. *Sensors, 18*, 1559. doi:10.3390/s18051559.

- [80] Ha, V., Inkpen, K. M., Mandryk, R. L., & Whalen, T. (2006). Direct intentions: The effects of input devices on collaboration around a tabletop display. *Proceedings of the 1<sup>st</sup> IEEE International Workshop on Horizontal Interactive Human Computer Systems (TABLETOP '06)*, 177–184, Adelaide, Australia: IEEE Computer Society. doi:10.1109/TABLETOP.2006.10.
- [81] Hornecker, E., Marshall, P. Dalton, N., & Rogers, Y. (2008). Collaboration and interference: Awareness with mice or touch input. In *Proceedings of the 2008 ACM Conference on Computer Supported Cooperative Work*, 167–176, San Diego, CA: Association for Computing Machinery. doi:10.1145/1460563.1460589.
- [82] Pinelle, D., Nacenta, M. A., Gutwin, C., & Stach, T. (2008). The effects of co-present embodiments on awareness and collaboration in tabletop groupware. *Proceedings of Graphics Interface 2008 (GI '08)*. Canadian Information Processing Society, 1–8.
- [83] Birnholtz, J. P., Grossman, T., Mak, C., & Balakrishnan, R. (2007). An exploratory study of input configuration and group process in a negotiation task using a large display. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '07)*. Association for Computing Machinery, New York, NY, 91–100. doi:10.1145/1240624.1240638.
- [84] Vogt, K., Bradel, L., Andrews, C., North, C., Endert, A., & Hutchings, D. (2011). Co-located collaborative sensemaking on a large high-resolution display with multiple input devices. In Campus, P., Graham, N., Jorge, J., Nunes, N., Palanque, P., & Winckler, M. (Eds.) *Human-Computer Interaction (INTERACT '11), Lecture Notes in Computer Science, vol. 6947*, 589–604. Berlin, Germany: Springer.
- [85] Nacenta, M. A., Pinelle, D., Stuckel, D., & Gutwin, C. (2007). The effects of interaction technique on coordination in tabletop groupware. *Proceedings of Graphics Interface 2007 (GI '07)*. Association for Computing Machinery, New York, NY, 191–198. doi:10.1145/1268517.1268550.
- [86] Muller-Tomfelde, C. (2009). Investigating temporal-spatial characteristics of mouse and touch input. In Gross, T. et al. (Eds.) *Human-Computer Interaction – Interact 2009*. Lecture Notes in Computer Science, vol 5726. Berlin: Springer. doi:10.1007/978-3-642-03655-2 72.
- [87] Shaer, O., Strait, M., Valdes, C., Feng, T., Lintz, M., & Wang, H. (2011). Enhancing genomic learning through tabletop interaction. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '11)*. Association for Computing Machinery, New York, NY, 2817–2826. doi:10.1145/1978942.1979361.
- [88] Alvarez, C., Brown, C., & Nussbaum, M. (2011). Comparative study of netbooks and tablet PCs for fostering face-to-face collaborative learning. *Computers in Human Behavior*, 27, 834–844. doi:10.1016/j.chb.2010.11.008.
- [89] Müller-Tomfelde, C. (2012). Interacting with mouse and touch devices on horizontal interactive displays. *Universal Access in the Information Society, 11*, 285–294. doi:10.1007/s10209-011-0238-8a.
- [90] Marshall, P., Hornecker, E., Morris, R., Dalton, S., & Rogers, Y. (2008). When the fingers do the talking: A study of group participation with varying constraints to a tabletop interface. *Proceedings the 3<sup>rd</sup> IEEE International Workshop on Horizontal Interactive Human Computer Systems*, 33–40. doi:10.1109/TABLETOP.2008.4660181.
- [91] Hinrichs, U., Hancock, M., Collins, C., & Carpendale, S. (2007). Examination of text-entry methods for tabletop displays. *Proceedings of the 2<sup>nd</sup> IEEE International Workshop on Horizontal Interactive Human Computer Systems (TABLETOP '07)*, 105–112, Newport, RI: IEEE Computer Society. doi:10.1109/TABLETOP.2007.10.

- [92] Rabin, E., & Gordon, A. M. (2004). Tactile feedback contributes to consistency of finger movements during typing. *Experimental Brain Research*, *155*(3), 362–369. doi:10.1007/s00221-003-1736-6.
- [93] Gunawardana, A., Paek, T., & Meek, C. (2010). Usability guided key-target resizing for soft keyboards. Proceedings of the 15<sup>th</sup> International Conference on Intelligent User Interfaces (IUI '10), 111–118, Hong Kong, China: ACM. doi:10.1145/1719970.1719986.
- [94] Findlater, L., & Wobbrock, J. O. (2012). Personalized input: Improving ten-finger touchscreen typing through automatic adaptation. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, 815–824, Austin, TX: Association for Computing Machinery. doi:10.1145/2207676.2208520.
- [95] Sax, C., Lau, H., & Lawrence, E. M. (2011). Liquid Keyboard: An ergonomic, adaptive QWERTY keyboard for touchscreens and surfaces. 5<sup>th</sup> International Conference on Digital Society (ICDS2011), 117–122. <u>http://hdl.handle.net/10453/16246</u> (accessed June 10, 2022).
- [96] Hoggan, E., Brewster, S. A., & Johnston, J. (2008). Investigating the effectiveness of tactile feedback for mobile touchscreens. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems* (CHI 2008), 1573–1582, Florence, Italy: Association for Computing Machinery. doi:10.1145/1357054.1357300.
- [97] Han, B., & Kim, K. (2015). Typing performance evaluation with multimodal soft keyboard completely integrated in commercial mobile devices. *Journal on Multimodal User Interfaces*, 9, 173–181. doi:10.1007/s12193-015-0177-4.
- [98] Fuccella, V., De Rosa, M., & Costagliola, G. (2014). Novice and expert performance of KeyScratch: A gesture-based text entry method for touch-screens. *IEEE Transactions on Human-Machine Systems*, *44*, 511–523. doi:10.1109/THMS.2014.2314447.
- [99] Mackenzie, I. S., & Soukoreff, W. (2002). Text entry for mobile computing: Models and methods, theory, and practice. *Human-computer interaction*, *17*, 147–198. doi:10.1080/07370024.2002.9667313.
- [100] Prattichizzo, D., Meli, L., & Malvezzi, M. (2015). Digital handwriting with a finger or a stylus: A biomechanical comparison. *IEEE Transactions on Haptics*, 8, 356–370. doi:10.1109/TOH.2015.2434812.
- [101] Ruan, S., Wobbrock, J. O., Liou, K., Ng, A., & Landay, J. A. (2017). Comparing speech and keyboard text entry for short messages in two languages on touchscreen phones. In *Proceedings of the ACM on Interactive, Mobile, Wearable, and Ubiquitous Technologies, 1*(4), Article 159. Association for Computing Machinery, New York, NY. doi:10.1145/3161187.
- [102] Martin, P., Tseu, A., Fe'rey, N., Touraine, D., & Bourdot, P. (2014). A hardware and software architecture to deal with multimodal and collaborative interactions in multiuser virtual reality environments. *Proceedings of SPIE 9012, The Engineering Reality of Virtual Reality 2014*, Article 901209. doi:10.1117/12.2042499.
- [103] Ch'ng, E. (2012). New ways of accessing information spaces using 3D multitouch tables. 2012 International Conference on Cyberworlds, 144–150. doi:10.1109/CW.2012.27.
- [104] Bruder, G., Steinicke, F., & Stuerzlinger, W. (2013). Touching the void revisited: Analyses of touch behaviour on and above tabletop surfaces. In Kotzé, P. et al. (Eds.) *Human-Computer Interaction* (INTERACT 2013). Lecture Notes in Computer Science, 8117, 278–296. Berlin, Germany: Springer.

- [105] Mendes, D., Fonesca, F., Araujo, B., Ferreira, A., & Jorge, J. (2014). Mid-air interactions above stereoscopic interactive tables. *IEEE Symposium on 3D User Interfaces (3DUI)*, 3–10. doi:10.1109/3DUI.2014.6798833.
- [106] Teather, R. J., & Stuerzlinger, W. (2011). Pointing at 3D targets in a stereo head-tracked virtual environment. *IEEE Symposium on 3D User Interfaces (3DUI 2011)*, 87–94. doi:10.1109/3DUI.2011.5759222.
- [107] Noor, A. K., & Aras, R. (2015). Potential of multimodal and multiuser interaction with virtual holography. *Advances in Engineering Software*, *81*, 1–6. doi:10.1016/j.advengsoft.2014.10.004.
- [108] Yuan, H., Calic, J., & Kondoz, A. (2014). Facilitating interaction with stereoscopic 3D display devices. 2014 3DTV Conference: The True Vision – Capture, Transmission, and Display for 3D video (3DTVCON), 1–4. doi:10.1109/3DTV.2014.6874770.
- [109] Oyekoya, O., Sassard, E., & Johnson, T. (2017). VizSpace: Interaction in the positive parallax screen plane. 2017 IEEE Symposium on 3D User Interfaces (3Dui), 229–230. doi:10.1109/3DUI.2017.7893355.
- [110] Xie, Y., Wang, D., & Sun, F. (2017). How visual discomfort changes with horizontal viewing angle on stereoscopic display. (Cornell University Technical Report 1811:08639, arXiv.org). <u>https://arxiv.org/ftp/arxiv/papers/1811/1811.08639.pdf</u> (accessed January 16, 2020).
- [111] Kulik, A., Kunert, A., Beck, S., Reichel, R., Blach, R., Zink, A., & Froehlich, B. (2011). C1x6: A stereoscopic six-user display for co-located collaboration in shared virtual environments. ACM Transactions on Graphics, 6, 1–12. doi:10.1145/2070781.2024222.
- [112] Wei, Y., Guan, D., Wang, Q., Li, X., Bian, Y., Qin, P.,... & Yang, C. (2019). Virtual fire drill system supporting co-located collaboration. *Virtual Reality & Intelligent Hardware*, 1, 290–302. doi:10.3724/SP.J.2096-5796.2019.0012.
- [113] Steinicke, F., Bruder, G., Hinrichs, K. H., Ropinski, T., & Lopes, M. (2008). 3D user interfaces for collaborative work. In Pavlidis, I. (Ed.) *Human Computer Interaction*. London, UK: IntechOpen. doi:10.5772/6297.
- [114] Surman, P., Sexton, I., Hopf, K., Lee, W. K., Neumann, F., Buckley, E., ... & Talukdar, S. (2012). Laser-based multi-user 3-D display. *Journal of the Society for Information Display*, 16, 743–753. doi:10.1889/1.2953481.
- [115] Kim, H., Lee, G. A., Yang, U., Kwak, T., & Kim, K.-H. (2012). Dual autostereoscopic display platform for multi-user collaboration with natural interaction. *ETRI Journal*, 34, 466–469. doi:10.4218/etrij.12.0211.0331.
- [116] Gutenko, I., Mirhosseini, S., & Kaufman, A. E. (2017). Angle and pressure-based volumetric picking on touchscreen devices. *Proceedings of the IEEE Symposium on 3D User Interfaces (3DUI 2017)*, 235–236. doi:10.1109/3DUI.2017.7893358.
- [117] Van de Camp, F., Hoppe, A., Unmussig, G., & Peinsipp-Byma, E. (2020). Cooperative and location-independent terrain assessment for deployment planning using a 3D mixed reality environment. *Proceedings of SPIE 11426, Virtual, Augmented, and Mixed Reality (XR) Technology for Multi-Domain Operations*, article 1142604. doi:10.1117/12.2558593.
- [118] Roberts, D., Wolff, R., Otto, O., & Steed, A. (2003). Constructing a gazebo: Supporting teamwork in a tightly coupled, distributed task in virtual reality. Presence, 12(6), 644–657. doi:10.1162/105474603322955932.

- [119] Lee, G., Kang, H., Lee, J., & Han, J. (2020). A user study on view-sharing techniques for one-to-many mixed reality collaborations. 2020 IEEE Conference on Virtual Reality and 3D User Interfaces, 343–352. doi:10.1109/VR46266.2020.00054.
- [120] Kiyokawa, K., Billinghurst, M., Hayes, S. E., Gupta, A., Sannohe, Y., & Kato, H. (2002). Communication behaviors of co-located users in collaborative AR interfaces. *Proceedings of the IEEE International Symposium on Mixed and Augmented Reality (ISMAR'02)*, 135–144, Darmstadt, Germany. doi:10.1109/ISMAR.2002.1115083.
- [121] Wolfartsberger, J., Zenizek, J., & Wild, N. (2020). Supporting teamwork in industrial virtual reality applications. *Procedia Manufacturing*, 42, 2–7. doi:10.1016/j.promfg.2020.02.016.
- [122] Mai, C., Bartsch, S. A., & Rieger, L. (2018). Evaluating shared surfaces for co-located mixed-presence collaboration. In *Proceedings of the 17th International Conference on Mobile and Ubiquitous Multimedia* (MUM 2018), 1–5. New York, NY: Association for Computing Machinery. doi:10.1145/3282894.3282910.
- [123] Marks, S., Windsor, J., & Wunsche, B. (2012). Head tracking based avatar control for virtual environment teamwork training. JVRB – Journal of Virtual Reality and Broadcasting, 9. doi:10.20385/1860-2037/9.2012.9.
- [124] Wang, L., Wu, W., Zhou, Z., & Popescu, V. (2020). View splicing for effective VR collaboration. 2020 IEEE International Symposium on Mixed and Augmented Reality (ISMAR), 509–519. doi:10.1109/ISMAR50242.2020.00079.
- [125] Lee, B., Hu, X., Cordeil, M., Prouzeau, A., Jenny, B., & Dwyer, T. (2021). Shared surfaces and spaces: Collaborative data visualisation in a co-located immersive environment. *IEEE Transactions on Visualization & Computer Graphics*, 27, 1171–1181. doi:10.1109/TVCG.2020.3030450.
- [126] Grandi, J. G., Debarba, H. G., & Maciel, A. (2019). Characterizing asymmetric collaborative interactions in virtual and augmented realities. 2019 IEEE Conference on Virtual Reality and 3D User Interfaces (VR), 127–135. doi:10.1109/VR.2019.8798080.
- [127] Gugenheimer, J., Stemasov, E., Frommel, J., & Rukzio, E. (2017). ShareVR: Enabling co-located experiences for virtual reality between HMD and non-HMD users. In *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems (CHI '17)*. Association for Computing Machinery, New York, NY, 4021–4033. doi:10.1145/3025453.3025683.
- [128] Emerson, L., Lipinski, R., Shirey, H., Malloy, T., & Marrinan, T. (2021). Enabling collaborative interaction with 360° panoramas between large-scale displays and immersive headsets. 2021 IEEE International Symposium on Mixed and Augmented Reality Adjunct (ISMAR-Adjunct), 183–188, doi:10.1109/ISMAR-Adjunct54149.2021.00045.
- [129] Olin, P. A., Issa, A. M., Feuchtner, T., & Gronbaek, K. (2020). 32<sup>nd</sup> Australian Conference on Human-Computer Interaction (OzCHI'20), 112–127. doi:10.1145/3441000.3441070.
- [130] Ouverson, K. M., & Gilbert, S. B. (2021). A composite framework of co-located asymmetric virtual reality. *Proceedings of the ACM on Human-Computer Interaction, 5*, Article 5. doi:10.1145/3449079.
- [131] Welsford-Ackroyd, F., Chalmers, A., Kuffner dos Anjos, R., Medeiros, D., Kim, H., & Rhee, T. (2021). Spectator View: Enabling asymmetric interaction between HMD Wearers and spectators with a large *display. Proceedings of the ACM on Human-Computer Interaction, 5*, Article 485. doi:10.1145/3486951.

- [132] Serubugo, S., Skantarova, D., Evers, N., Kraus, M. (2018). Facilitating asymmetric collaborative navigation in room-scale virtual reality for public spaces. In Brooks, A., Brooks, E., & Vidakis, N. (Eds.) *Interactivity, Game Creation, Design, Learning, and Innovation*. Lecture Notes of the Institute for Computer Sciences, Social Informatics and Telecommunications Engineering, vol. 229. Chamonix, Switzerland: Springer. doi:10.1007/978-3-319-76908-0 7.
- [133] Bork, F., Lehner, A., Kugelmann, D., Eck, U., Waschke, J., & Navab, N. (2019). VesARlius: An augmented reality system for large-group co-located anatomy learning. 2019 IEEE International Symposium on Mixed and Augmented Reality Adjunct (ISMAR-Adjunct), 122–123. doi:10.1109/ISMAR-Adjunct.2019.00-66.
- [134] Sikorski, B., Leoncini, P., & Luongo, C. (2020). A glasses-based holographic tabletop for collaborative monitoring of aerial missions. In De Paolis, L., & Bourdot, P. (Eds.), *Augmented Reality, Virtual Reality, and Computer Graphics (AVR 2020)*. Lecture Notes in Computer Science, vol. 12242. Chamonix, Switzerland: Springer. doi:10.1007/978-3-030-58465-8 26.
- [135] Nijholt, A. (2022). Capturing obstructed nonverbal cues in augmented reality interactions: A short survey. In Bhaumik, S., Chattopadhyay, S., Chattopadhyay, T., & Bhattacharya, S. (Eds.), *Proceedings of International Conference on Industrial Instrumentation and Control*. Lecture Notes in Electrical Engineering, vol. 815. Singapore: Springer. doi:10.1007/978-981-16-7011-4 1.
- [136] Huang, L., Collins, S., Kobayashi, L., Merck, D., & Sgouros, T. (2019). Shared visualizations and guided procedure simulation in augmented reality with Microsoft HoloLens. In *Proceedings of Medical Imaging* 2019: Image-Guided Procedures, Robotic Interventions, and Modeling, vol. 10951, p. 1095112, San Diego, CA: International Society for Optics and Photonics. doi:10.1117/12.2511321.
- [137] Butscher, S., Hubenschmid, S., Müller, J., Fuchs, J., & Reiterer, H. (2018). Clusters, trends, and outliers: How immersive technologies can facilitate the collaborative analysis of multidimensional data. *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems (CHI '18)*, Paper 90, Montreal, QC: Association for Computing Machinery. doi:10.1145/3173574.3173664.
- [138] Chen, L., Liu, Y., Yu, L., Gao, B. Y., Caon, M., Yue, Y., & Liang, H.-N. (2021). Effect of visual cues on pointing tasks in co-located augmented reality collaboration. *Symposium on Spatial User Interaction* (SUI'21), Article 12. doi:10.1145/3485279.3485297.
- [139] Aromaa, S., Väätänen, A., Aaltonen, I., Goriachev, V., Helin, K., & Karjalainen, J. (2020). Awareness of the real-world environment when using augmented reality head-mounted display. *Applied Ergonomics*, 88, doi:10.1016/j.apergo.2020.103145.
- [140] Lukosch, S., Billinghurst, M., Alem, L., & Kiyokawa, K. (2015). Collaboration in augmented reality. Computer Supported Cooperative Work (CSCW), 24(6), 515–525. doi:10.1007/s10606-015-9239-0.
- [141] Radu, I., Vivek, H., & Schneider, B. (2020). Unequal impacts of augmented reality on learning and collaboration during robot programming with peers. *Proceedings of the ACM on Human-Computer Interaction, 4 (CSCW3), Article 245. doi:10.1145/3432944.*
- [142] Jansen, P., Fischbach, F., Gugenheimer, J., Stemasov, E., Frommel, J., & Rukzio, E. (2020). ShARe: Enabling co-located asymmetric multi-user interaction for augmented reality head-mounted displays. *Proceedings of the 33rd Annual ACM Symposium on User Interface Software and Technology (UIST '20)*. Association for Computing Machinery, New York, NY, 459–471. doi:10.1145/3379337.3415843.

- [143] Rowen, A., Grabowski, M., Rancy, J.-P., & Crane, A. (2019). Impacts of wearable augmented reality displays on operator performance, situation awareness, and communication in safety-critical systems. *Applied Ergonomics*, 80, 17–27. doi:10.1016/j.apergo.2019.04.013.
- [144] Nilsson, S., Johansson, B., & Jonsson, A. (2009). Using AR to support cross-organisational collaboration in dynamic tasks. *Proceedings of the δ<sup>th</sup> IEEE International Symposium on Mixed and Augmented Reality*, 3–12, Orlando, FL: IEEE Computer Society. doi:10.1109/ISMAR.2009.5336522.
- [145] Croft, B. L., Lucero, C., Neurnberger, D., Greene, F., Qiu, A., Higgins, R., & Gustafson, E. (2018). Command and control collaboration sand table (C2-CST). In Chen, J., & Fragomeni, G. (Eds.) Virtual, Augmented and Mixed Reality: Applications in Health, Cultural Heritage, and Industry (VAMR 2018). Lecture Notes in Computer Science, 10910. Chamonix, Switzerland: Springer. doi:10.1007/978-3-319-91584-5\_20.
- [146] Su, S., Perry, V., Roy, H., Gamble, K., & Kase, S. (2020). 3D user interface for a multi-user augmented reality mission planning application. In Cassenti, D. (Ed.) Advances in Human Factors and Simulation, AHFE 2019. Advances in Intelligent Systems and Computing, vol. 958. Chamonix, Switzerland: Springer. doi:10.1007/978-3-030-20148-7 12.
- [147] Ens, B., Goodwin, S., Prouzeau, A., Anderson, F., Wang, F. Y., Gratzl, S., Lucarelli, Z., ..., & Dwyer, T. (2021). Uplift: A Tangible and immersive tabletop system for casual collaborative visual analytics. *IEEE Transactions on Visualization and Computer Graphics*, 27, 1193–1203. doi:10.1109/TVCG.2020.3030334.
- [148] Dedual, N. J., Oda, O., & Feiner, S. K. (2011). Creating hybrid user interfaces with a 2D multi-touch tabletop and a 3D see-through head-worn display. *Proceedings of the 10<sup>th</sup> IEEE International Symposium* on Mixed and Augmented Reality, 231–232, Basel, Switzerland: IEEE Computer Society. doi:10.1109/ISMAR.2011.6092391.
- [149] Zhang, L., Chen, S., Dong, H., & El Saddik, A. (2018). Visualizing Toronto city data with HoloLens. *IEEE Consumer Electronics Magazine*, 7(3), 73–80. doi:10.1109/MCE.2018.2797658.
- [150] Seichter, H. (2004). Benchworks: Augmented reality in urban design. Proceedings of the 9<sup>th</sup> International Conference on Computer-Aided Architectural Design Research in Asia (CAADRIA 2004), 937–946, Seoul, South Korea.
- [151] Plasson, C., Cunin, D., Laurillau, Y., & Nigay, L. (2019). Tabletop AR with HMD and Tablet: A comparative study for 3D selection. *Proceedings of the 2019 ACM International Conference on Interactive Surfaces and Spaces (ISS '19)*, 409–414, New York, NY: Association for Computing Machinery. doi:10.1145/3343055.3360760.
- [152] Morrison, A., Oulasvirta, A., Peltonen, P., Lemmela, S., Jacucci, G., Reitmayr, G., et al. (2009). Like bees around the hive: A comparative study of a mobile augmented reality map. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '09)*, 1889–1898, Boston, MA: Association for Computing Machinery. doi:10.1145/1518701.1518991.
- [153] Touel, S., Mekkadem, M., Kenoui, M., & Benbelkacem, S. (2017). Collocated learning experience within collaborative augmented environment (anatomy course). *5th International Conference on Electrical Engineering - Boumerdes (ICEE-B)*, 1–5. doi:10.1109/ICEE-B.2017.8192219.
- [154] Wells, T., & Houben, S. (2020). CollabAR Investigating the mediating role of mobile AR interfaces on co-located group collaboration. *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems (CHI '20)*. Association for Computing Machinery, New York, NY, 1–13. doi:10.1145/3313831.3376541.

#### DRDC-RDDC-2022-R139

- [155] De Belen, R. A. J., Nguyen, H., Filonik, D., Del Favero, D., & Bednarz, T. (2019). A systematic review of the current state of collaborative mixed reality technologies: 2013-2018. *AIMS Electronics & Electrical Engineering*, 3(2), 181–223. doi:10.3934/ElectrEng.2019.2.181.
- [156] Ens, B., Lanir, J., Tang, A., Bateman, S., Lee, G., Piumsomboon, T., & Billinghurst, M. (2019). Revisiting collaboration through mixed reality: The evolution of groupware. *International Journal of Human-Computer Studies*, 131, 81–91. doi:10.1016/j.ijhcs.2019.05.011.
- [157] Isenberg, T. (2016). Interactive exploration of three-dimensional scientific visualizations on large display surfaces. In Anslow, C., Campos, P., & Jorge, J. (Eds.) *Collaboration Meets Interactive Spaces*. Chamonix, Switzerland: Springer. doi:10.1007/978-3-319-45853-3 6
- [158] Alexander, T., Helge, R., & Conradi, J. (2006). Applicability of virtual environments as C4ISR displays. In *Virtual Media for Military Applications* (pp. 1-1 – 1-12). Meeting Proceedings RTO-MP-HRM-136, paper 1.
- [159] Khundam, C., Vorachart, V., Preeyawongsakul, P., Hosap, W., & Noel, F. (2021). A comparative study of interaction time and usability of using controllers and hand tracking in virtual reality training. Informatics, 8, Article 60. doi:10.3390/informatics8030060.
- [160] Hinckley, K., Pausch, R., Goble, J. C., & Kassell, N. F. (1994). A survey of design issues in spatial input. Proceedings of the 7<sup>th</sup> annual ACM Symposium on User Interface Software and Technology (UIST '94), 213–222, Marina Del Ray, CA: Association for Computing Machinery. doi:10.1145/192426.192501.
- [161] Epps, J., Lichman, S., & Wu, M. (2006). A study of hand shape use in tabletop gesture interaction. In CHI '06 Extended Abstracts on Human Factors in Computing Systems (CHI EA 2006), 748–753. Association for Computing Machinery, New York, NY. doi:10.1145/1125451.1125601.
- [162] Piumsomboon, T., Clark, A., Billinghurst, M., & Cockburn, A. (2013). User-defined gestures for augmented reality. *CHI 2013 Extended Abstracts on Human Factors in Computing Systems (CHI EA '13)*, 955–960, Paris, France: Association for Computing Machinery. doi:10.1145/2468356.2468527.
- [163] Yim, D., Loisonm, G. N., Fard, F. H., Chan, E., McAllister, A., & Maurer, F. (2016). Gesture-driven interactions on a virtual hologram in mixed reality. *Proceedings of the 2016 ACM Companion on Interactive Surfaces and Spaces (ISS '16 Companion)*, 55–61, Niagara Falls, ON: Association for Computing Machinery. doi:10.1145/3009939.3009948.
- [164] Dactu, D., Lukosch, S., & Brazier, F. (2015). On the usability and effectiveness of different interaction types in augmented reality. *International Journal of Human-Computer Interaction*, 31(3), 193–209. doi:10.1080/10447318.2014.994193.
- [165] Buschel, W., Mitschick, A., Meyer, T., & Dachselt, R. (2019). Investigating smartphone-based pan and zoom in 3D data spaces in augmented reality. *Proceedings of the 21<sup>st</sup> International Conference on Human-Computer Interaction with Mobile Devices and Services (MobileHCI '19)*. Association for Computer Machinery, New York, NY, Article 2, 1–13. doi:10.1145/3338286.3340113.
- [166] Sereno, M., Besancon, L., & Isenberg, T. (2019). Supporting volumetric data visualization and analysis by combining augmented reality visuals with multi-touch input. *EuroVis Posters*, June 2019, Porto, Portugal, 21–23. doi:10.2312/eurp.20191136.
- [167] Ericson, M. A., Bolia, R. S., Nelson, T., & McKinley, R. L. (1999). Operational constraints on the utility of virtual audio cueing. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 43, 1206–1209. doi:10.1177/154193129904302208.

- [168] Wallace, J. R., & Scott, S. D. (2009). Investigating teamwork and taskwork in single- and multi-display groupware systems. *Personal and Ubiquitous Computing*, *13*, 569–581. doi:10.1007/s00779-009-0241-8.
- [169] Niu, S., McCrickard, D. S., Nguyen, J., Haqq, D., Kotut, L., Stelter, T. L., & Fox, E. A. (2020). Investigating paradigms of group territory in multiple display environments. *Proceedings of the ACM on Human-Computer Interaction*, 4, GROUP, Article 13, 28 pages. doi:10.1145/3375193.
- [170] Gutwin, C., & Greenberg, S. (1998). Design for individuals, design for groups: tradeoffs between power and workspace awareness. *Proceedings of the 1998 ACM Conference on Computer Supported Cooperative Work (CWCW '98)*, 207–216, Seattle, WA: Association for Computing Machinery. doi:10.1145/289444.289495.
- [171] Bolstad, C. A., & Endsley M. R. (2000). The effect of task load and shared displays on team situation awareness. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 44, 189–192. doi:10.1177/154193120004400150.
- [172] Liu, J., Qin, Y., Yan, Q., Yu, C., & Shi, Y. (2015). A tabletop-centric smart space for emergency response. *IEEE Pervasive Computing*, *14*, 32–40. doi:10.1109/MPRV.2015.24.
- [173] Langner, R., Kister, U., & Dachselt, R. (2019). Multiple coordinated views at large displays for multiple users: Empirical findings on user behavior, movements, and distances. *IEEE Transactions of Visualization and Computer Graphics*, 25, 608–618. doi:10.1109/TVCG.2018.2865235.
- [174] Sinmai, K., & Andras, P. (2014). Mapping on surfaces: Supporting collaborative work using interactive tabletop. In Baloian N., Burstein F., Ogata H., Santoro F., & Zurita G. (Eds.), *Collaboration and Technology (CRIWG 2014)*. Lecture Notes in Computer Science, vol 8658. Chamonix, Switzerland: Springer. doi:10.1007/978-3-319-10166-8\_29.
- [175] Cheng, K., He, L., Meng, X., Shamma, D. A., Nguyen, D., & Thangapalam, A. (2015). CozyMaps: Real-time collaboration on a shared map with multiple displays. *Proceedings of the 17th International Conference on Human-Computer Interaction with Mobile Devices and Services (MobileHCI '15)*, 46–51. doi:10.1145/2785830.2785851.
- [176] Seifert, J., Simeone, A., Schmidt, D., Holleis, P., Reinartz, C., Wagner, M., ... & Rukzio, E. (2012). MobiSurf: Improving co-located collaboration through integrating mobile devices and interactive surfaces. In *Proceedings of the 2012 ACM international conference on Interactive tabletops and surfaces* (*ITS '12*). New York, NY: Association for Computing Machinery, 51–60. doi:10.1145/2396636.2396644.
- [177] Chegini, M., Andrews, K., Schreck, T., & Sourin, A. (2019). Multiple linked-view exploration on large displays facilitated by a secondary handheld device, *Proceedings of SPIE 11049, International Workshop* on Advanced Image Technology (IWAIT), Article 110490H. doi:10.1117/12.2521648.
- [178] Tong, L., Tabard, A., George, S., & Serna, A. (2017). Horizontal vs. Vertical: How the orientation of a large interactive surface impacts collaboration in multi-surface environments. In Bernhaupt, R., Dalvi, G., Joshi, A., K. Balkrishan, D., O'Neill, J., Winckler, M. (Eds.) *Human-Computer Interaction – INTERACT* 2017. INTERACT 2017. Lecture Notes in Computer Science, vol 10515. Chamonix, Swizerland: Springer. doi:10.1007/978-3-319-67687-6\_14.
- [179] Xiong, Z., Huang, C., Wang, H., Liu, X., Huang, X., & Jiang, L. (2021). Evaluation of writing efficiency when using tablet computers at inclines. *IEEE Access*, 9, 144149–144156. doi:10.1109/ACCESS.2021.3121202.

- [180] Martinez-Maldonado, R., Clayphan, A., Yacef, K., & Kay, J. (2015). MTFeedback: Providing notifications to enhance teacher awareness of small group work in the classroom. *IEEE Transactions on Learning Technologies*, 8, 187–200. doi:10.1109/TLT.2014.2365027.
- [181] Bachl, S., Tomitsch, M., Kappel, K., & Grechenig, T. (2011). The effects of personal displays and transfer techniques on collaboration strategies in multi-touch based multi-display environments. In Campos, P., Grahamm, N., Jorge, J., Nunes, N., Palanque, P., & Winckler, M. (Eds.) *Human-Computer Interaction* (INTERACT 2011). Lecture Notes in Computer Science, vol. 6948. Berlin: Springer. doi:10.1007/978-3-642-23765-2\_26.
- [182] Chung, H., North, C., Zeitz Self, J., Chu, S., & Quek, F. (2014). VisPorter: Facilitating information sharing for collaborative sensemaking on multiple displays. *Personal and Ubiquitous Computing*, 18, 1169–1186. doi:10.1007/s00779-013-0727-2.
- [183] Waldner, M., & Schmalstieg, D. (2011). Collaborative information linking: Bridging knowledge gaps between users by linking across applications. 2011 IEEE Pacific Visualization Symposium, 115–122. doi:10.1109/PACIFICVIS.2011.5742380.
- [184] Craig, P., & Liu, Y. (2018). Coordinating user selections in collaborative smart-phone large-display multi-device environments. In Luo, Y. (Ed.) *Cooperative Design, Visualization, and Engineering*. CDVE 2018. Lecture Notes in Computer Science, 11151. Chamonix, Switzerland: Springer. doi:10.1007/978-3-030-00560-3\_4.
- [185] Cheng, K., He, L., Meng, X., Shamma, D. A., Nguyen, D., & Thangapalam, A. (2015). CozyMaps: Real-time collaboration on a shared map with multiple displays. In *Proceedings of the 17<sup>th</sup> International Conference on Human-Computer Interaction with Mobile Devices and Services*, 46–51.
- [186] Han, J., Bae, S. H., & Suk, H.-J. (2017). Visual discomfort and visual fatigue: Comparing head-mounted display and smartphones. *Journal of the Ergonomic Society of Korea*, 36, 293–303. doi:10.5143/JESK.2017.36.4.293.
- [187] Guo, J., Weng, D., Fang, H., Zhang, Z., Ping, J., Liu, Y., & Wang, Y. (2020). Exploring the differences of visual discomfort caused by long-term immersion between virtual environments and physical environments. 2020 IEEE Conference on Virtual Reality and 3D User Interfaces (VR), 443–452, doi:10.1109/VR46266.2020.00065.
- [188] Choy, S.-M., Cheng, E., Wilkinson, R. H., Burnett, I., & Austin, M. W. (2021). Quality of experience comparison of stereoscopic 3D videos in different projection devices: Flat screen, panoramic screen and virtual reality headset. *IEEE Access*, 9, 9584–9594. doi:10.1109/ACCESS.2021.3049798.
- [189] Hirota, M., Kanda, H., Endo, T., Miyoshi, T., Miyagawa, S., Hirohara, Y., ... & Fujikado, T. (2019). Comparison of visual fatigue caused by head-mounted display for virtual reality and two-dimensional display using objective and subjective evaluation. *Ergonomics*, 62, 759–766. doi:10.1080/00140139.2019.1582805.
- [190] Howarth, P. A. (2011). Potential hazards of viewing 3-D stereoscopic television, cinema and computer games: a review. *Ophthalmic & Physiological Optics*, *31*, 111–122. doi:10.1111/j.1475-1313.2011.00822.
- [191] Hoffman, D. M., Girshick, A. R., Akeley, K., & Banks, M. S. (2008). Vergence-accommodation conflicts hinder visual performance and cause visual fatigue. *Journal of Vision*, *8*, Article 33. doi:10.1167/8.3.33.
- [192] Kramida, G. (2016). Resolving the vergence-accommodation conflict in head-mounted displays. *IEEE Transactions on Visualization and Computer Graphics*, 22, 1912–1931. doi:10.1109/TVCG.2015.2473855.

- [193] Guo, J., Weng, D., Duh, H. B.-L., Liu, Y., & Wang, Y. (2017). Effects of using HMDs on visual fatigue in virtual environments. 2017 IEEE Virtual Reality (VR), 249–250. doi:10.1109/VR.2017.7892270.
- [194] Iskander, J., Hossny, M., & Nahavandi, S. (2019). Using biomechanics to investigate the effect of VR on eye vergence system. *Applied Ergonomics*, *81*, Article 102883. doi:10.1016/j.apergo.2019.102883.
- [195] Ballestin, G., Solari, F., & Chessa, M. (2018). Perception and action in peripersonal space: A comparison between video and optical see-through augmented reality devices. 2018 IEEE International Symposium on Mixed and Augmented Reality Adjunct (ISMAR-Adjunct), 184–189. doi:10.1109/ISMAR-Adjunct.2018.00063.
- [196] Drouot, M., Le Bigot, N., Bolloc'h, J., Bricard, E., de Bougrenet, J.-L., & Nourrit, V. (2021). The visual impact of augmented reality during an assembly task. Displays, 66, 101987. doi:10.1016/j.displa.2021.101987.
- [197] Marklin, R. W. Jr., Toll, A. M., Bauman, E. H., Simmins, J. J., LaDisa Jr, J. F., & Cooper, R. (2020). Do head-mounted augmented reality devices affect muscle activity and eye strain of utility workers who do procedural work? Studies of operators and manhole workers. *Human Factors*, 64, 305–323. doi:10.1177/0018720820943710.
- [198] Bando, T., Iijima, A., & Yano, S. (2012). Visual fatigue caused by stereoscopic images and the search for the requirement to prevent them: A review. *Displays*, *33*, 76–83. doi:10.1016/j.displa.2011.09.001.
- [199] Gabbard, J. L., Mehra, D. G., & Swan, J. E. (2019). Effects of AR display context switching and focal distance switching on human performance. *IEEE Transactions on Visualization and Computer Graphics*, 25, 2228–2241. doi:10.1109/TVCG.2018.2832633.
- [200] Microsoft Corporation. *Comfort Mixed reality* | *Microsoft Docs*. June 25, 2020. https://docs.microsoft.com/en-us/windows/mixed-reality/design/comfort (accessed July 12, 2020).
- [201] Arefin, M. S., Phillips, N., Plopski, A., Gabbard, J. L., & Swan, J. E. (2022). The effect of context switching, focal switching distance, binocular and monocular viewing, and transient focal blur on human performance in optical see-through augmented reality. *IEEE Transactions on Visualization and Computer Graphics*, 28, 2014–2025. doi:10.1109/TVCG.2022.3150503.
- [202] Yu, J., & Kim, G. J. (2015). Eye strain from switching focus in optical see-through displays. In Abascal, J., Barbosa, S., Fetter, M., Gross, T., Palanque, P., & Winckler, M. (Eds.) *Human-Computer Interaction – INTERACT 2015*. Lecture Notes in Computer Science, vol. 9299. Chamonix, Switzerland: Springer. doi:10.1007/978-3-319-22723-8 59.
- [203] Fang, H., Weng D., Guo, J., Shen, R., Jiang, H., & Tu, Z. (2020). Potential effects of dynamic parallax on eyesight in virtual reality system. 2020 IEEE Conference on Virtual Reality and 3D User Interfaces Abstracts and Workshops (VRW), 784–785, doi:10.1109/VRW50115.2020.00243.
- [204] Leroy, L., Fuchs, P., & Moreau, G. (2012). Visual fatigue reduction for immersive stereoscopic displays by disparity, content, and focus-point adapted blur. *IEEE Transactions on Industrial Electronics*, 59, 3998–4004. doi:10.1109/TIE.2011.2171173.
- [205] Erickson, A., Kim, K., Bruder, G., & Welch, G. F. (2020). Effects of dark mode graphics on visual acuity and fatigue with virtual reality head-mounted displays. 2020 IEEE Conference on Virtual Reality and 3D User Interfaces (VR), 434–442, doi:10.1109/VR46266.2020.00064.

- [206] Kruijff, E., Swan, J. E. II, & Feiner, S. (2010). Perceptual issues in augmented reality revisited. Proceedings of the 2010 IEEE International Symposium on Mixed and Augmented Reality, 3–12, Seoul, South Korea: IEEE Computer Society. doi:10.1109/ISMAR.2010.5643530.
- [207] Kooi, F. L., Toet, A., & Hoving, S. (2019). Visual processing of symbology in head-fixed large field-of-view displays. *Journal of Vision, 19*, Article 85b. doi:10.1167/19.10.85b.
- [208] Hirzle, T., Fischbach, F., Karlbauer, J., Jansen, P., Gugenheimer, J., Rukzio, E., & Bulling, A. (2022). Understanding, addressing, and analysing digital eye strain in virtual reality head-mounted displays. ACM Transactions on Computer-Human Interaction, 29, Article 33, 1–80. doi:10.1145/3492802.
- [209] Porcino, T. M., Clua, E. W., Vasconcelos, C. N., Trevisan, D., & Valente, L. (2016). Minimizing cyber sickness in head mounted display systems: Design guidelines and applications. *IEEE 5<sup>th</sup> International Conference on Serious Games and Applications for Health (SeGAH 2017).* <u>https://arxiv.org/pdf/1611.06292.pdf</u> (accessed July 21, 2020).
- [210] Porcino, T., Trevisan, D., & Clua, E. (2020). Minimizing cybersickness in head-mounted display systems: Causes and strategies review. 22nd Symposium on Virtual and Augmented Reality (SVR), 154–163. doi:10.1109/SVR51698.2020.00035.
- [211] Barrett, J. (2004). Side effects of virtual environments: A review of the literature. Defence Science and Technology Organization, Canberra, Australia. Document No. DSTO-TR-1419.
- [212] North Atlantic Treaty Organization Research and Technology Organization. (2000). The capability of virtual reality to meet military requirements. RTO-MP-054.
- [213] Szpak, A., Michalski, S. C., Saredakis, D., Chen, C. S., & Loetscher, T. (2019). Beyond feeling sick: The visual and cognitive aftereffects of virtual reality. *IEEE Access*, 7, 130883–130892. doi:10.1109/ACCESS.2019.2940073.
- [214] Mittelstaedt, J. M., Wacker, J., & Stelling, D. (2019). VR aftereffect and the relation of cybersickness and cognitive performance. *Virtual Reality*, *23*, 143–154. doi:10.1007/s10055-018-0370-3.
- [215] Kolasinski, E. M. (1995). Simulator sickness in virtual environments. Technical Report, U.S. Army Research Institute for the Behavioral and Social Sciences (ARI-1027).
- [216] Stauffert, J.-P., Niebling, F., & Lotoschik, M. E. (2020). Latency and cybersickness: Impact, causes, and measures. A review. *Frontiers in Virtual Reality*, *1*. doi:10.3389/frvir.2020.582204.
- [217] Kwok, K. K., Ng, A. D. T., & Lau, H. Y. K. (2018). Effect of navigation speed and VR devices on cybersickness. 2018 IEEE International Symposium on Mixed and Augmented Reality Adjunct (ISMAR-Adjunct), 91–92. doi:10.1109/ISMAR-Adjunct.2018.00041.
- [218] Kim, J., Luu, W., & Palmisano, S. (2020). Multisensory integration and the experience of scene instability, presence, and cybersickness in virtual reality environments. *Computers in Human Behavior*, 113, Article 106484. doi:10.1016/j.chb.2020.106484.
- [219] Deok, P. W., Wook, J. S., Ho, K. Y., Ah, K. G., Ho, S. W., & Sang, K. Y. (2017). A study on cyber sickness reduction by oculo-motor exercise performed immediately prior to viewing virtual reality (VR) content on head mounted display (HMD). *Vibroengineering PROCEDIA*, 14, 260–264. doi:10.21595/vp.2017.19170.

- [220] Kroma, A., Kassm, N. A., & Teather, R. J. (2021). Motion sickness conditioning to reduce cybersickness. *Symposium on Spatial User Interaction (SUI'21)*, Article 22, 1–2. doi:10.1145/3485279.3485300.
- [221] Andre, L., & Coutellier, R. (2019). Cybersickness evaluation of a remediation system: A pilot study. 2019 International Conference on 3D Immersion (IC3D), 1–6. doi:10.1109/IC3D48390.2019.8975989.
- [222] Farmani, Y., & Teather, R. J. (2020). Evaluating discrete viewpoint control to reduce cybersickness in virtual reality. *Virtual Reality*, *24*, 645–664. doi:10.1007/s10055-020-00425-x.
- [223] Chang, E., Kim, H. T., & Yoo, B. (2021). Predicting cybersickness based on user's gaze behaviors in HMD-based virtual reality. *Journal of Computational Design & Engineering*, 8, 728–739. doi:10.1093/jcde/qwab010.
- [224] Lin, Y.-X., Venkatakrishnan, R., Venkatakrishnan, R., Ebrahimi, E., Lin, W.-C., & Babu, S. V. (2020). How the presence and size of static peripheral blur affects cybersickness in virtual reality. *ACM Transactions on Applied Perception*, 17, Article 16. doi:10.1145/3419984.
- [225] Risi, D., & Palmisano, S. (2019). Effects of postural stability, active control, exposure duration and repeated exposures on HMD induced cybersickness. *Displays*, *60*, 9–17. doi:10.1016/j.displa.2019.08.003.
- [226] Dumanska, N., Strojny, P., & Strojny, A. (2018) Can simulator sickness be avoided? A review on temporal aspects of simulator sickness. *Frontiers in Psychology*, 9. doi:10.3389/fpsyg.2018.02132.
- [227] Tian, N., Lopes, P., & Boulic, R. (2022). A review of cybersickness in head-mounted displays: Raising attention to individual susceptibility. *Virtual Reality*, online first publication. doi:10.1007/s10055-022-00638-2.
- [228] Soyka, F., Kokkinara, E., Leyrer, M., Buelthoff, H., Slater, M., & Mohler, B. (2015). Turbulent motions cannot shake VR. *2015 IEEE Virtual Reality (VR)*, 33–40. doi:10.1109/VR.2015.7223321.
- [229] Ng, A. K. T., Chan, L. K. Y., & Lau, H. Y. K. (2020). A study of cybersickness and sensory conflict theory using a motion-coupled virtual reality system. *Displays*, 61, Article 101922. doi:10.1016/j.displa.2019.08.004.
- [230] Stevens, A. H., & Butkiewicz, T. (2019). Reducing seasickness in onboard marine VR use through visual compensation of vessel motion. 2019 IEEE Conference on Virtual Reality and 3D User Interfaces (VR), 1872–1873. doi:10.1109/VR.2019.8797800.
- [231] McGill, M., Ng, A., & Brewster, S. (2017). I am the passenger: How motion cues can influence sickness for in-car VR. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI 2017)*, 5655–5668, Denver, CO: Association for Computing Machinery. doi:10.1145/3025453.3026046.
- [232] Pettijohn, K. A., Peltier, C., Lukos, J. R., Norris, J. N., & Biggs, A. T. (2020). Virtual and augmented reality in a simulated naval engagement: Preliminary comparisons of simulator sickness and human performance. *Applied Ergonomics*, 89, Article 103200. doi:10.1016/j.apergo.2020.103200.
- [233] Patterson, F., & Muth, E. (2010). Cybersickness onset with reflexive head movements during land and shipboard head-mounted display flight simulation. Naval Aerospace Medical Research Laboratory, NAMRL-10-43.
- [234] Park, S.-H., & Lee, G.-C. (2020). Full-immersion virtual reality: Adverse effects related to static balance. *Neuroscience Letters*, 733, Article 134974. doi:10.1016/j.neulet.2020.134974.

#### DRDC-RDDC-2022-R139

- [235] Oh, H.-J., & Lee, G.-C. (2021). Feasibility of full immersive virtual reality video game on balance and cybersickness of healthy adolescents. *Neuroscience Letters*, 760, Article 136063. doi:10.1016/j.neulet.2021.136063.
- [236] Vovk, A., Wild, F., Guest, W., & Kuula, T. (2018). Simulator sickness in augmented reality training using the Microsoft HoloLens. *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems (CHI '18)*, Paper 209, Montreal, QC: Association for Computing Machinery. doi:10.1145/3173574.3173783.
- [237] Maraj, C., Hurter, J., & Murphy, S. (2020). Performance, simulator sickness, and immersion of a ball-sorting task in virtual and augmented realities. *In Virtual, Augmented and Mixed Reality. Design and Interaction: 12th International Conference (VAMR 2020), Held as Part of the 22nd HCI International Conference* (HCII 2020), Proceedings, Part I. Berlin: Springer-Verlag, 522–534. doi:10.1007/978-3-030-49695-1\_35.
- [238] Allcoat, D., Hatchard, T., Azmat, F., Stansfield, K., Watson, D., & von Mühlenen, A. (2021). Education in the digital age: Learning experience in virtual and mixed realities. *Journal of Educational Computing Research*, 59, 795–816. doi:10.1177/0735633120985120.
- [239] Hughes, C. L., Fidopiastis, C., Stanney, K. M., Bailey, P. S., & Ruiz, E. (2020). The psychometrics of cybersickness in augmented reality. *Frontiers in Virtual Reality*, *1*. doi:10.3389/frvir.2020.602954.
- [240] Livingston, M. A., Ai, Z., & Decker, J. W. (2009). A user study towards understanding stereo perception in head-worn augmented reality displays. *Proceedings of the 8<sup>th</sup> IEEE International Symposium on Mixed and Augmented Reality*, 53–56, Orlando, FL: IEEE Computer Society. doi:10.1109/ISMAR.2009.5336496.
- [241] McIntire, J. P., Havig, P. R., Harrington, L. K., Wright, S. T., Watamaniuk, S. N. J., & Heft, E. (2018). Microstereopsis is good, but orthostereopsis is better: Precision alignment task performance and viewer discomfort with a stereoscopic 3D display. *Proceedings of the SPIE Volume 10666: Three-Dimensional Imaging, Visualization, and Display 2018*, Article 106660Q. doi:10.1117/12.2297378.
- [242] Metcalfe, J. S., Davis, J. A. Jr., Tauson, R. A., & McDowell, K. (2008). Assessing constraints on soldier cognitive and perceptual motor performance during vehicle motion. U.S. Army Research Laboratory: Human Research & Engineering Directorate, ARL-TR-4661.
- [243] Hunter, A., Bourgeois, N., & Langlois, R. (2014). The impact of motion induced interruptions on cognitive performance. *Proceedings of the 5<sup>th</sup> International Conference on Applied Human Factors and Ergonomics (AHFE2014)*, 3481–3487, Krakow, Poland. Defence Research and Development Canada, External Literature, DRDC-RDDC-2014-P82.
- [244] Valk, P. J. L., Grech, M. R., & Bos, J. E. (2010). A multi-factorial analysis of human performance during a 9-day sea trial. In *HPAS 2010 International Conference On Human Performance at Sea*, Glasgow, Scotland, United Kingdom.
- [245] Duncan, C., Hickey, C. J., & Byrne, J. M. (2018). The effects of a moving environment on postural control and task performance during manual materials handling, visual tracking, and arithmetic tasks. *International Journal of Industrial Ergonomics*, 66, 221–229. doi:10.1016/j.ergon.2018.03.007.
- [246] Cockburn, A., Gutwin, C., Palanque, P., Deleris, Y., Trask, C., Coveney, A., Yung, M., & MacLean, K. (2017). Turbulent touch: Touchscreen input for cockpit flight displays. *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems*, 6742–6753, Denver, CO: Association for Computing Machinery. doi:10.1145/3025453.3025584.

- [247] McLeod, P., Poulton, C., Du Ross, H., & Lewis, W. (1980). The influence of ship motion on manual control skills. *Ergonomics*, *23*(7), 623–634. doi:10.1080/00140138008924777.
- [248] Lin, C. J., Liu, C. N., Chao, C. J., & Chen, H. J. (2010). The performance of computer input devices in a vibration environment. *Ergonomics*, *53*(4), 478–490. doi:10.1080/00140130903528186.
- [249] Bjørneseth, F. B., Dunlop, M. D., & Hornecker, E. (2012). Assessing the effectiveness of direct gesture interaction for a safety critical marine application. *International Journal of Human-Computer Studies*, 70(10), 729–745. doi:10.1016/j.ijhcs.2012.06.001.
- [250] Yau, Y.-J., Chao, C.-J., Feng, W.-Y., & Hwang, S.-L. (2011). The effects of platform motion and target orientation on the performance of trackball manipulation. *Ergonomics*, 54, 745–754. doi:10.1080/00140139.2011.595512.
- [251] Wynne, R. A., Parnell, K. J., Smith, M. A., Plant, K. L., & Stanton, N. A. (2021). Can't touch this: Hammer time on touchscreen task performance variability under simulated turbulent flight conditions. *International Journal of Human-Computer Interaction*, 37,666–679. doi:10.1080/10447318.2021.1890492.
- [252] Dodd, S. R., Lancaster, J., Groethe, S., DeMers, B., Rogers, B., & Miranda, A. (2014). Touch on the flight deck: The impact of display location, size, touch technology, & turbulence on pilot performance. *Proceedings of the 2014 IEEE/AIAA 33<sup>rd</sup> Digital Avionics Systems Conference (DASC)*, Paper 2C3-1, Colorado Springs, CO: IEEE Computer Society. doi:10.1109/DASC.2014.6979570.
- [253] Cockburn, A., Masson, D., Gutwin, C., Palanque, P., Goguey, A., Yung, M., Cris, C., & Trask, D. (2019). Design and evaluation of braced touch for touchscreen input stabilisation. *International Journal of Human-Computer Studies*, 122, 21–37. doi:10.1016/j.ijhcs.2018.08.005.
- [254] Salmon, P. M., Lenne, M. G., Triggs, T., Goode, N., Conelissen, M., & Demczuk, V. (2011). The effects of motion on in-vehicle touch screen system operation: A battle management system case study. *Transportation Research Part F: Traffic Psychology and Behaviour, 14*, 494–503. doi:10.1016/j.trf.2011.08.002.
- [255] Nakashima, A., & Cheung, B. (2006). The effects of vibration frequencies on physical, perceptual and cognitive performance. DRDC Toronto, Technical Report, DRDC-TORONTO-TR-2006-218.
- [256] Dobie, T. G. (2003). Critical significance of human factors in ship design. Proceedings of the 2003 RVOC Meeting, October 8–10, 2003. Large Lakes Observatory, University of Minnesota.
- [257] Tao, D., Zeng, J., Shen, D., Qu, X., & Zhang, T. (2021). Touchscreen operations in vibration conditions: Task precision requirement matters. In 2020 2nd International Conference on E-Business and E-commerce Engineering (EBEE 2020). Association for Computing Machinery, New York, NY, USA, 60–64. doi:10.1145/3446922.3446933.
- [258] Conway, G. E., Szalma, J. L., & Hancock, P. A. (2007). A quantitative meta-analytic examination of whole-body vibration effects on human performance. *Ergonomics*, 50(2), 228–245. doi:10.1080/00140130600980888.
- [259] Wang, H., Tao, D., Cai, J., & Qu, X. (2021). Effects of vibration and target size on the use of varied computer input devices in basic human-computer interaction tasks. *Human Factors & Ergonomics in Manufacturing & Service Industries*, 32, 199–213. doi:10.1002/hfm.20938.

- [260] Nicholson, G. (2011). Head-mounted display (HMD) assessment for tracked vehicles. Proceedings of SPIE 8042: Display Technologies and Applications for Defense, Security, and Avionics V; and Enhanced and Synthetic Vision 2011, Paper 80420I, Orlando, FL: SPIE Electronic Imaging. doi:10.1117/12.887102.
- [261] Garza, J. L., & Young, J. G. (2015). A literature review of the effects of computer input devices design on biomechanical loading and musculoskeletal outcomes during computer work. *Work*, 52, 217–230. doi:10.3233/WOR-152161.
- [262] Radwan, A., Kallasy, T., Monroe, A., Chrisman, E., & Carpenter, O. (2018). Benefits of alternative computer mouse designs: A systematic review of controlled trials. *Cogent Engineering*, 5, Article 1521503. doi:10.1080/23311916.2018.1521503.
- [263] Hedge, A., Feathers, D., & Rollings, K. (2010). Ergonomic comparison of slanted and vertical computer mouse designs. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 54, 561–565. doi:10.1177/154193121005400604.
- [264] Quemelo, P. R. V., & Vieira, E. R. (2013). Biomechanics and performance when using a standard and vertical computer mouse. *Ergonomics*, *56*, 1336–1344. doi:10.1080/00140139.2013.805251.
- [265] Houwink, A., Oude Hengel, K. M., Odell, D., & Dennerlein, J. T. (2009). Providing training enhances the biomechanical improvements of an alternative mouse design. *Human Factors*, 51, 46–55. doi:10.1177/0018720808329843.
- [266] Burgess-Limerick, R., & Green, B. (2000). Using multiple case studies in ergonomics: An example of pointing device use. *International Journal of Industrial Ergonomics*, 26, 381–388. doi:10.1016/S0169-8141(00)00013-5.
- [267] Onyebeke, L. C., Young, J. G., Trudeau, M. B., & Dennerlein, J. T. (2014). Effects of forearm and palm supports on the upper extremity during computer mouse use. *Applied Ergonomics*, 45, 564–570. doi:10.1016/j.apergo.2013.07.016.
- [268] Cook, C., Burgess-Limerick, R., & Papalia, S. (2004). The effect of upper extremity support on upper extremity posture and muscle activity during keyboard use. *Applied Ergonomics*, 35, 285–292. doi:10.1016/j.apergo.2003.12.005.
- [269] Muller, C., Tomatis, L., & Laubli, T. (2010). Muscular load and performance compared between a pen and a computer mouse as input devices. *International Journal of Industrial Ergonomics*, 40, 607–617. doi:10.1016/j.ergon.2010.08.004.
- [270] Alcaraz-Mateos, E., Hernández-Gómez, R., Calvente, E. R., Sánchez-Campoy, N., González-Moro, I. M., Caballero-Alemán, F., & Poblet, E. (2022). Comparison of muscle activity while using different input devices in digital pathology. *Revista Española de Patología*, 55, 19–25. doi:10.1016/j.patol.2021.02.005.
- [271] Hemati, K., Mirjalili, S., Ghasemi, M. S., Abdolahian, Y., Siroos, R., Pardis, S., Aghilinejad, M., & Deghan, N. (2020). Functional parameters, wrist posture deviations and comfort: A comparison between a computer mouse and a touch input pen as devices. *Work*, 65. doi:10.3233/WOR-203124.
- [272] Kim, J. H., Aulck, L., Bartha, M. C., Harper, C. A., & Johnson, P. W. (2014). Differences in typing forces, muscle activity, comfort, and typing performance among virtual, notebook, and desktop keyboards. *Applied Ergonomics*, 45, 1406–1413. doi:10.1016/j.apergo.2014.04.001.

- [273] Bachynskyi, M., Palmas, G., Oulasvirta, A., Steimle, J., & Weinkauf, T. (2015). Performance and ergonomics of touch surfaces: A comparative study using biomechanical simulation. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI 2015)*, 1817–1826, Seoul, South Korea: Association for Computing Machinery. doi:10.1145/2702123.2702607.
- [274] Chourasia, A. O., Wiegmann, D. A., Chen, K. B., Irwin, C. B., & Sesto, M. E. (2013). Effect of sitting or standing on touch screen performance and touch characteristics. *Human Factors*, 55, 789–802. doi:10.1177/0018720812470843.
- [275] Al-Megren, S., Kharrufa, A., Hook, J., Holden, A., Sutton, S., & Olivier, P. (2015). Comparing fatigue when using large horizontal and vertical multi-touch interaction displays. In Abascal, J. et al. (Eds.) *Human-Computer Interaction (INTERACT 2015). Lecture Notes in Computer Science, 9299*, 156–164. Chamonix, Switzerland: Springer.
- [276] Jeong, H., & Liu, Y. (2017). Effects of touchscreen gesture's type and direction on finger-touch input performance and subjective ratings. *Ergonomics*, *60*, 1528–1539. doi:10.1080/00140139.2017.1313457.
- [277] Kang, H., & Shin, G. (2017). Effects of touch target location on performance and physical demands of computer touchscreen use. *Applied Ergonomics*, *61*, 159–167. doi:10.1016/j.apergo.2017.01.015.
- [278] Davis, K. G., Hammer, M. J., Kotowski, S. E., & Bhattacharya, A. (2015). An ergonomic comparison of data entry work using a keyboard vs. touch screen input device while standing and sitting. *Journal of Ergonomics, S4*, Article 007. doi:10.4172/2165-7556.S4-007.
- [279] Kang, H., & Shin, G. (2014). Hand usage pattern and upper body discomfort of desktop touchscreen users. *Ergonomics*, *9*, 1397–1404. doi:10.1080/00140139.2014.924574.
- [280] Asakawa, D. S., Becker, M. G., Asaro, J. M., & Hein, J. L. (2019). Shoulder, elbow, and wrist joint angle excursions vary by gesture during touchscreen interaction. *Journal of Electromyography and Kinesiology*, Article 102377. doi:10.1016/j.jelekin.2019.102377.
- [281] Mayer, S., Gad, P., Wolf, K., Wozniak, P. W., & Henze, N. (2017). Understanding the ergonomic constraints in designing for touch surfaces. *Proceedings of the 19th International Conference on Human-Computer Interaction with Mobile Devices and Services (MobileHCI '17)*, Article 33, 1–9. doi:10.1145/3098279.3098537.
- [282] Li, B., Zhang, W., Zhou, R., Yang, C., & Li, Z. (2012). A comparative ergonomics study: Performing reading-based tasks on large-scale tabletop vs. laptop. *International Journal of Industrial Ergonomics*, 42, 156–161. doi:10.1016/j.ergon.2011.10.003.
- [283] Odeghe, O., & Udosen, U. J. (2013). Design and manufacture of an ergonomic computer pen. *Advanced Materials Research*, 824, 239–245. doi:10.4028/www.scientific.net/AMR.824.239.
- [284] Hincapié-Ramos, J. D., Guo, X., Moghadasian, P., & Irani, P. (2014). Consumed endurance: A metric to quantify arm fatigue of mid-air interactions. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '14)*, 1063–1072, Toronto, ON: Association for Computing Machinery. doi:10.1145/2556288.2557130.
- [285] Satriadi, K. A., Ens, B., Cordeil, M., Jenny, B., Czauderna, T., & Willett, W. (2019). Augmented reality map navigation with freehand gestures. *Proceedings of the 2019 IEEE Conference on Virtual Reality and* 3D User Interfaces, 593–603, Osaka, Japan: IEEE Computer Society. doi:10.1109/VR.2019.8798340.

- [286] Harrison, C., Ramamurthy, S., & Hudson, S. E. (2012). On-body interaction: Armed and dangerous. Proceedings of the 6<sup>th</sup> International Conference on Tangible, Embedded, and Embodied Interaction (TEI '12). Association for Computing Machinery, New York, NY, 69–76. doi:10.1145/2148131.2148148.
- [287] Bampatzia, S., Antoniou, A., Lepouras, G., Roumelioti, E., & Bravou, V. (2015). Comparing game input modalities: A study for the evaluation of player experience by measuring emotional state and game usability. *Proceedings of the IEEE 9<sup>th</sup> International Conference on Research Challenges in Information Science (RCIS 2015)*, 530–531. doi:10.1109/RCIS.2015.7128922.
- [288] Ahmed, S., Leroy, L., & Bouaniche, A. (2017). Questioning the use of virtual reality in the assessment of the physical impacts of real-task gestures and tasks. 23<sup>rd</sup> International Conference on Virtual System & Multimedia (VSMM 2017), 1–10. doi:10.1109/VSMM.2017.8346271.
- [289] Wentzel, J., d'Eon, G., & Vogel, D. (2020). Improving virtual reality ergonomics through reach-bounded non-linear input amplification. *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*. Association for Computing Machinery, New York, NY, 1–12. doi:10.1145/3313831.3376687.
- [290] Kim, E., & Shin, G. (2018). Head rotation and muscle activity when conducting document editing tasks with a head-mounted display. *Proceedings of the Human Factors & Ergonomics Society Annual Meeting*, 62, 952–955. doi:10.1177/1541931218621219.
- [291] Theis, S., Alexander, T., Mayer, M., & Wille, M. (2013). Considering ergonomic aspects of head-mounted displays for applications in industrial manufacturing. In Duffy V.G. (Ed.) Digital Human Modeling and Applications in Health, Safety, Ergonomics, and Risk Management. Human Body Modeling and Ergonomics. DHM 2013. *Lecture Notes in Computer Science, vol 8026*. Springer, Berlin. doi:10.1007/978-3-642-39182-8\_34.
- [292] Garrett, G. A., Reid, C. R., Jenkins, M., Talbot, T., & Doherty, S. (2018). Effects of prolonged use of mixed reality systems in occupational settings. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 62, 2104–2106. doi:10.1177/1541931218621474.
- [293] Cometti, C., Païzis, C., Casteleira, A., Pons, G., & Babault, N. (2018). Effects of mixed reality head-mounted glasses during 90 minutes of mental and manual tasks on cognitive and physiological functions. *PeerJ*, 6, e5847. doi:10.7717/peerj.5847.
- [294] Kim, E., & Shin, G. (2021). User discomfort while using a virtual reality headset as a personal viewing system for text-intensive office tasks. *Ergonomics*, *64*, 891–899. doi:10.1080/00140139.2020.1869320.
- [295] Ito, K., Tada, M., Ujike, H., & Hyodo, K. (2021). Effects of the weight and balance of head-mounted displays on physical load. *Applied Sciences*, *11*, Article 6802. doi:10.3390/app11156802.
- [296] Nguyen, N. Q., Cardinell, J., Ramjist, J. M., Androutsos, D., & Yang, V. X. D. (2020). Augmented reality and human factors regarding the neurosurgical operating room workflow. *Optical Architectures for Displays and Sensing in Augmented, Virtual, and Mixed Reality (AR, VR, MR), 11310*, 113100D. doi:10.1117/12.2546483.
- [297] Van Benthem, K. V., Cobert, H., & Zobarich, R. (2021). Human factors and ergonomics considerations when using augmented reality head mounted displays—Literature analysis report. Defence Research and Development Canada, Contract Report, DRDC-RDDC-2021-C198.
- [298] Wille, M., & Wischniewski, S. (2015). Influence of head mounted display hardware on performance and strain. *Proceedings of the Human Factors and Ergonomics Society Europe*, 269–279.

- [299] Lozano, C., Jindrich, D., & Kahol, K. (2011). The impact on musculoskeletal system during multitouch tablet interactions. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems* (CHI '11). Association for Computing Machinery, New York, NY, 825–828. doi:10.1145/1978942.1979062.
- [300] Young, J. G., Trudeau, M. Odell, D., Marinelli, K., & Dennerlein, J. T. (2012). Touch-screen tablet user configurations and case-supported tilt affect head and neck flexion angles. *Work*, 41(1), 81–91. doi:10.3233/WOR-2012-1337.
- [301] Young, J. G., Trudeau, M. B., Odell, D., Marinelli, K., & Dennerlein, J. T. (2013). Wrist and shoulder posture and muscle activity during touch-screen tablet use: Effects of usage configuration, tablet type, and interacting hand. *Work*, 45(1), 59–71. doi:10.3233/WOR-131604.
- [302] Dennerlein, J. T. (2015). The state of ergonomics for mobile computing technology. *WORK: A* Journal of Prevention, Assessment, & Rehabilitation, 52, 269–277. doi:10.3233/WOR-152159.
- [303] Livingston, M. A., Rosenblum, J. L., Julier, S. J., Brown, D., Baillot, Y., Swan, J. E. II, Gabbard, J. L., & Hix, D. (2002). An augmented reality system for military operations in urban terrain. *Proceedings of the Interservice/Industry Training, Simulation, & Education Conference (I/ITSEC '02)*, Orlando, FL.
- [304] Livingston, M. A. (2005). Evaluating human factors in augmented reality systems. *IEEE Computer Graphics and Applications*, 26(6), 6–9. doi:10.1109/MCG.2005.130.
- [305] LaViola, J., Williamson, B., Brooks, C., Veazanchin, S., Sottilare, R., & Garrity, P. (2015). Using augmented reality to tutor military tasks in the wild. *Proceedings of the Interservice/Industry Training, Simulation, & Education Conference (I/ITSEC '15)*, Orlando, FL.
- [306] Chuter, A. (2018, November 26). BAE rolls out augmented-reality suite for the Royal Navy. *Defense News*. <u>https://www.defensenews.com/</u> (accessed January 16, 2020).
- [307] Benko, H., Ishak, E. W., & Feiner, S. (2004). Collaborative mixed reality visualization of an archaeological excavation. *Proceedings of the 3<sup>rd</sup> IEEE and ACM International Symposium on Mixed and Augmented Reality*, 132–140, Arlington, VA: IEEE Computer Society. doi:10.1109/ISMAR.2004.23.
- [308] Caputo, F., De Amicis, R., Di Gironomo, G., & Stork, A. (2003). Ergonomic driven design in augmented reality. *Proceedings of XIII ADM XV INGEGRAF International Conference on Tools and Methods Evolution in Engineering Design*, Naples, Italy.
- [309] Wafaa, A. M., Bonnefoy, N. D., Dubois, E., Torguet, P., & Jessel, J. (2008). Virtual reality simulation for prototyping augmented reality. *Proceedings of the 2008 International Symposium on Ubiquitous Virtual Reality*, 55–58, Gwangju, South Korea: IEEE Computer Society. doi:10.1109/ISUVR.2008.9.
- [310] Steves, M. P., Morse, E., Gutwin, C., & Greenberg, S (2001). A comparison of usage evaluation and inspection methods for assessing groupware usability. *Proceedings of the 2001 International ACM SIGGROUP Conference on Supporting Group Work (GROUP '01)*, 125–134, Boulder, CO: Association for Computing Machinery. doi:10.1145/500286.500306.
- [311] Pinelle, D., & Gutwin, C. (2008). Evaluating teamwork support in tabletop groupware applications using collaboration usability analysis. *Personal & Ubiquitous Computing*, 12(3), 237–254. doi:10.1007/s00779-007-0145-4.

- [312] Gutwin, C., & Greenberg, S. (2000). The mechanics of collaboration: Developing low cost usability evaluation methods for shared workspaces. *Proceedings of the IEEE 9<sup>th</sup> International Workshop on Enabling Technologies: Infrastructure for Collaborative Enterprises (WET-ICE '00)*, 98–103, Gaithersberg, MD: IEEE Computer Society. doi:10.5555/647068.715651.
- [313] Hart, S. G., & Staveland, L. E. (1988). Development of NASA-TLX (Task Load Index): Results of empirical and theoretical research. In P. A. Hancock and N. Meshkati (Eds.) *Human Mental Workload*. Amsterdam: North Holland Press.
- [314] Poole, A., & Ball, L. J. (2006). Eye tracking in HCI and usability research. In Ghaoui, C. (Ed.), *Encyclopedia of Human Computer Interaction*, 211–219. IGI Global. doi:10.4018/978-1-59140-562-7.ch034.
- [315] Fuller, R. M., & Dennis, A. R. (2009). Does fit matter? The impact of task-technology fit and appropriation on team performance in repeated tasks. *Information Systems Research*, 20(1), 2–17.
- [316] Goodhue, D., & Thompson, R. (1995). Task-technology fit and individual performance. *Management Information Systems Quarterly*, 19(2), 213–236. doi:10.2307/249689.
- [317] Howard, M. C., & Rose, J. C. (2019). Refining and extending task-technology fit theory: Creation of two task-technology fit scales and empirical clarification of the construct. *Information & Management*, 56(6), 103134. doi:10.1016/j.im.2018.12.002.
- [318] Gribbins, M., Subramaniam, C., & Shaw, M. (2006). Process-Technology Fit: Extending Task-Technology Fit to assess enterprise information technologies. *Proceedings of the 27th International Conference on Information Systems*, 61–70, Milwaukee, WI: ICIS.
- [319] Endsley, T. C., Sprehn, K. A., Brill, R. M., Ryan, K. J., Vincent, E. C., & Martin, J. M. (2017). Augmented reality design heuristics: Designing for dynamic interactions. Augmented reality design heuristics: Designing for dynamic interactions. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 61, 2100–2104. doi:10.1177/1541931213602007.
- [320] Van der Muelen, J., & Smith, J. R. (2015). The effect of 2-dimensional and 3-dimensional perspective view displays on situation awareness during command and control. 2015 IEEE International Multidisciplinary Conference on Cognitive Methods in Situation Awareness and Decision, 89–95. doi:10.1109/COGSIMA.2015.7108180.

**Table A.1:** A summary of the findings from the integrative review, assessing the extent to which a given display or input technology (rows) satisfies the criteria of interest for CRACCEN (columns).

	Supports Individual Work	Supports Teamwork & Communication	Suitable to UWW Environment & Ergonomics	Comments
Central Display				
Desktop Display	High (baseline)	Low [33] [88] [155] Medium to high when combined with a group display [172]	Medium to High [273] [279] [282]	Ideal for individual work, currently employed widely on naval vessels. May be useful for the individual work aspects of mixed-focus collaboration.
Handheld Display	High [34] [35]	Low to Medium [34] [35] [88] [152] Medium to high when combined with a group display [172] [173] [174] [175] [176] [177] [178] [181]	Low to Medium [61] [179] [186] [252] [254] [273] [299] [300] [301] [302]	Good for individual work and okay for group work; but a number of ergonomic considerations. May be useful for the individual work aspects of mixed-focus collaboration.
Large Vertical Display	Medium [33] [38]	Low to Medium [33] [38] [39] [48] [49] [50] [52] [58] [178]	Medium [273] [274] [275] [278] [279]	Very good for leader-follower communications or static group displays, less ideal for equitable collaboration.
Large Tabletop Display	Medium [45] [46]	High [33] [34] [35] [38] [39] [40] [41] [42] [43] [44] [45] [46] [50] [51] [52] [53] [56] [58] [147] [178]	Medium [77] [273] [275] [278] [279] [282]	Ideal for collaboration and communication; some ergonomic and design considerations.

	Supports Individual Work	Supports Teamwork & Communication	Suitable to UWW Environment & Ergonomics	Comments	
2D Input Methods <sup>9</sup>					
Direct Touch	High [59] [60] [61] [62] [63] [64] [65] [66] [67] [68] [69] [70] [76] [77] [78] [100]	High [79] [80] [81] [82] [85] [86] [88] [89] [90]	Medium [66] [76] [77] [80] [86] [246] [247] [248] [249] [251] [252] [253] [254] [257] [259] [276] [277] [278] [279] [280] [281] [287]	Ideal for collaboration; some ergonomic and environmental considerations.	
Pen Styli	High [62] [69] [70] [71] [72] [100]	Medium [80]	Medium [80] [269] [270] [271] [283] (practical considerations) <sup>10</sup>	More precise than direct touch; may not be offset by pragmatic considerations for multiple pens/users but little research has looked at their use in collaborative tasks.	
2D Mouse	High [59] [60] [61] [62] [63] [64] [65] [67] [73] [74] [75] [262]	Low [38] [80] [81] [82] [83] [84] [86] [89] [90]	Medium [73] [77] [80] [86] [246] [247] [248] [249] [252] [253] [259] [261] [262] [263] [264] [265] [267] [269] [270] [271] [287]	Not ideal for use with a tabletop, or for collaboration. Some ergonomic and environmental considerations.	
Trackball	High [73] [75]	N/A	Medium to High [73] [246] [248] [250] [259] [266]	Not examined for teamwork, commonly used in shipboard environments.	
Physical Keyboard	High [67] [91] [92]	Low (practical considerations)	Medium to High [247] [272] (practical considerations)	Ideal speed/accuracy and ergonomics. Less practical for collaborative work due to the need for physical keyboards.	
Soft Keyboard	Medium to High [67] [91] [92] [93] [94] [96] [97] [98] [99]	Medium to High (practical considerations)	Medium to High [93] [95] [96] [97] [272]	Can be improved with gestures, haptic feedback, and dynamic resizing of keys. Multiple typists can work at once.	
Handwriting Recognition	Medium to High [91] [99] [100]	N/A	Medium to High [283]	Not examined for teamwork; some ergonomic considerations.	

<sup>&</sup>lt;sup>9</sup> The assessments for input methods depend heavily on the display technology used. The ratings shown here assume the large horizontal display as the central display. <sup>10</sup> Practical considerations denote additional issues that were discussed in the text but do not have references associated with them; these are typically common-sense or

self-evident physical constraints that would not necessarily have been researched.

	Supports Individual WorkSupports Teamwork & CommunicationSuitable to UWW Environment & Ergonomics		Comments		
Speech Recognition	Medium [91] [101]	Low [91] [102]	Low (practical considerations)	Noise, jargon, and acronyms make it impractical for UWW environment; not ideal for multiple users.	
3D Information					
Stereoscopic 3D Displays with Glasses	Low to Medium [103] [104] [105] [106] [107] [110]	Low (Medium*) [107] [111] [112]	Low [110] [190] [191]	Limited ability to interact with displays, not examined for teamwork. *Exception: Euclideon's multi-user glasses system	
Auto-stereoscopic and Other 3D Displays	Low to Medium [29] [30] [113] [116] [157]	Low to Medium [109] [114] [115]	Low [190] [191]	Limited ability to interact with displays, with exception of volumetric displays. Little research on teamwork.	
VR HMD	Medium to High [117] [125] [139] [159]	Low to Medium [117] [118] [120] [121] [122] [123] [124] [125] [126] [127] [128] [129] [130] [131] [139]	Low [186] [187] [188] [189] [191] [192] [193] [194] [195] [199] [205] [208] [211] [212] [213] [214] [215] [216] [217] [219] [220] [225] [226] [228] [229] [230] [231] [232] [233] [234] [235] [237] [238] [260] [286] [288] [289] [290] [291] [294] [295] (practical considerations)	Not ideal for co-located collaboration but may be good for remote collaboration. Ergonomic and environmental considerations. Cybersickness may be exacerbated in UWW environments.	
AR HMD	Medium to High [136] [137] [139] [144] [151] [160] [164] [195]	High [31] [120] [126] [133] [134] [136] [137] [138] [139] [140] [141] [142] [143] [144] [145] [146] [147] [155] [166]	Medium [151] [160] [164] [191] [192] [195] [196] [197] [199] [201] [202] [206] [232] [236] [237] [238] [239] [240] [241] [260] [284] [285] [286] [290] [291] [292] [293] [295] [297] [298]	Ideal for collaboration; particularly strong combination with tabletop display. Some ergonomic and environmental considerations. Cybersickness may be exacerbated in UWW environments.	
AR with handheld device	Medium to High [151] [154] [161] [162]	Medium [126] [129] [152] [153] [154]	Low to Medium [151] [273] [301]	Okay for collaboration, but a number of ergonomic considerations.	

A rating is given for each technology's fit to each criterion: Low, Medium, High, or N/A in the event that insufficient evidence was available in the literature to support a rating. Relevant references to support each rating are provided in the row below each technology.

# List of symbols/abbreviations/acronyms/initialisms

2D	two-dimensional
3D	three-dimensional
AI	Artificial Intelligence
AOO	Area of Operations
AR	Augmented Reality
AS	Auto-stereoscopic
ASPO	Anti-Submarine Plot Officer
ASWC	Anti-Submarine Warfare Commander
C2	Command and Control
CANDID	Canadian Defence Information Database
CAVEs	Cave Automatic Virtual Environments
СО	Commanding Officer
COpsO	Current Operations Officer
COA	Course of Action
CRACCEN	Command Reconnaissance Area Coordination and Control Environmental Network
DRDC	Defence Research and Development Canada
DTIC	Defense Technical Information Center
FOpsO	Future Operations Officer
FTC-SS	Force Track Coordinator Subsurface
HGA	Hierarchical Goal Analysis
HMD	Head Mounted Display
IEEE	Institute of Electrical and Electronics Engineers
ORO	Operations Room Officer
RCN	Royal Canadian Navy
SA	Situation Awareness
SCS	Sonar Control Supervisor
SME	Subject Matter Expert
TG	Task Group
TTF	Task Technology Fit
UWW	Underwater Warfare
UWWD	Underwater Warfare Director
VR	Virtual Reality
WBE	Work Breakdown Element

	<b>DOCUMENT CONTROL DATA</b> *Security markings for the title, authors, abstract and keywords must be entered when the document is sensitive					
1.	ORIGINATOR (Name and address of the organization preparing the document. A DRDC Centre sponsoring a contractor's report, or tasking agency, is entered in Section 8.)			<ul> <li>SECURITY MARKING (Overall security marking of the document including special supplemental markings if applicable.)</li> </ul>		
	DRDC – Atlantic Research Centre Defence Research and Development Canada 9 Grove Street		CAN UNCLASSIFIED			
	P.O. Box 1012		2b. CONTROLLED GOODS			
	Dartmouth, Nova Scotia B2Y 3Z7 Canada			NON-CONTROLLED GOODS DMC A		
3.	TITLE (The document title and sub-title as indicated on the title page.)					
	A review of display and input technologies for the development of the Command Reconnaissance Area Coordination and Control Environmental Network (CRACCEN) decision aid system					
4.	AUTHORS (Last name, followed by initials - ranks, titles, etc., not	to be used)				
	Chubala, C. M.					
5.	DATE OF PUBLICATION (Month and year of publication of document.)	<li>6a. NO. OF PAGES (Total pages, including Annexes, excluding DCD, covering and verso pages.)</li>		including cluding DCD,	6b. NO. OF REFS (Total references cited.)	
	August 2022		6	2	320	
7.	DOCUMENT CATEGORY (e.g., Scientific Report, Contract Report, Scientific Letter.)					
	Scientific Report					
8.	<ul> <li>SPONSORING CENTRE (The name and address of the department project office or laboratory sponsoring the research and development.)</li> <li>DRDC – Atlantic Research Centre</li> <li>Defence Research and Development Canada</li> <li>9 Grove Street</li> <li>P.O. Box 1012</li> <li>Dartmouth, Nova Scotia B2Y 3Z7</li> <li>Canada</li> </ul>					
9a.	PROJECT OR GRANT NO. (If appropriate, the applicable research and development project or grant number under which the document was written. Please specify whether project or grant.) CRACCEN	9b. CONTRACT NO. (If appropriate, the applicable number under which the document was written.)				
10a	DRDC PUBLICATION NUMBER (The official document number by which the document is identified by the originating activity. This number must be unique to this document.)10b. OTHER DOCUMENT NO(s). (Any other numbers which may be assigned this document either by the originator or by the sponsor.)					
	DRDC-RDDC-2022-R139					
11a	In a. FUTURE DISTRIBUTION WITHIN CANADA (Approval for further dissemination of the document. Security classification must also be considered.)					
	Public release					
11b	11b. FUTURE DISTRIBUTION OUTSIDE CANADA (Approval for further dissemination of the document. Security classification must also be considered.)					
12.	2. KEYWORDS, DESCRIPTORS or IDENTIFIERS (Use semi-colon as a delimiter.)					
	decision aid; tactical decision aids; underwater surveillance; underwater warfare; human factors					

13. ABSTRACT (When available in the document, the French version of the abstract must be included here.)

The Command Reconnaissance Area Coordination and Control Environmental Network (CRACCEN) is a decision aid system being developed by Defence Research and Development Canada (DRDC) – Atlantic Research Centre to revolutionize the way the Royal Canadian Navy (RCN) conducts Underwater Warfare (UWW). This Scientific Report outlines the high-level goals and requirements of the CRACCEN system and reviews several fields of research in order to determine the appropriate candidate technologies for which to develop the system. A variety of potential display and input hardware technologies are evaluated with regard to both task requirements and characteristics of the UWW task environment. The results of this review suggest that a tabletop display capable of accommodating a group of simultaneous users be adopted as the central display for the CRACCEN system, with the possibility of using additional displays and display modes to augment the capabilities of the UWW team. Further research with Subject Matter Experts (SMEs) is recommended to validate the viability, usability, and added value of these displays, and to iteratively design interfaces that are well-suited to both the display type and the needs of the CRACCEN users.

Le Centre de recherche de l'Atlantique de Recherche et développement pour la défense Canada (RDDC) élabore actuellement un réseau environnemental de coordination et de contrôle de la zone de reconnaissance du commandement (CRACCEN). Il s'agit d'un système d'aide à la décision visant à révolutionner la façon dont la Marine royale canadienne (MRC) mène la guerre sous-marine (GSM). Ce rapport scientifique énonce les objectifs et les exigences de haut niveau du système CRACCEN et examine plusieurs domaines de recherche afin de proposer des technologies envisageables qui conviennent à son élaboration. Le rapport évalue une variété d'écrans d'affichage et de technologies matérielles d'acquisition de données en ce qui a trait aux exigences opérationnelles de même qu'à l'environnement opérationnel de la GSM. Les résultats de cet examen indiguent que l'affichage central choisi du système CRACCEN pourrait être sous forme de surface de table. Cet affichage permet d'accueillir plusieurs utilisateurs simultanément et offre a possibilité d'utiliser des écrans et des modes d'affichages supplémentaires afin d'accroître les capacités de l'équipe de la GSM. Il est recommandé de poursuivre les recherches avec des experts pour vérifier la viabilité, la convivialité et la valeur ajoutée de ces écrans, ainsi que pour concevoir des interfaces de manière itérative qui conviennent au type d'écran d'affichage et aux besoins des utilisateurs du système CRACCEN.